UTMS 2008-20

July 24, 2008

Analysis and estimation of error constants for P_0 and P_1 interpolations over triangular finite elements

by

Xuefeng LIU and Fumio KIKUCHI



UNIVERSITY OF TOKYO

GRADUATE SCHOOL OF MATHEMATICAL SCIENCES KOMABA, TOKYO, JAPAN

Analysis and Estimation of Error Constants for P_0 and P_1 Interpolations over Triangular Finite Elements

Xuefeng LIU, Fumio KIKUCHI*

Graduate School of Mathematical Sciences, University of Tokyo 3-8-1 Komaba, Meguro, Tokyo, 153-8914 Japan

Abstract

We give some fundamental results on the error constants for the piecewise constant interpolation function and the piecewise linear one over triangles. For the piecewise linear one, we mainly analyze the conforming case, but the present results also appear to be available for the non-conforming case. We obtain explicit relations for the upper bounds of the constants, and analyze dependence of such constants on the geometric parameters of triangles. In particular, we explicitly determine some special constants including the Babuška-Aziz constant, which plays an essential role in the interpolation error estimation of the linear triangular finite element. The obtained results are expected to be widely used for a priori and a posteriori error estimations in adaptive computation and numerical verification of numerical solutions based on the triangular finite elements. We also give some numerical results for the error constants and for a posteriori estimates of some eigenvalues related to the error constants.

Keywords : FEM, error estimates, triangular finite elements, interpolation error constants. *Mathematical Subject Classification* 2000 : 65N15, 65N30

1 Introduction

The finite element method (FEM) is now recognized as a powerful numerical method for wide classes of partial differential equations. Furthermore, it also has sound mathematical bases such as highly refined a priori and a posteriori error estimations. In the classical a priori error analysis of FEM, the interpolation error analysis is essential to derive final error estimates in various norms and/or semi-norms [10, 11, 21]. In this process, there appear a number of positive constants besides the standard discretization (or mesh) parameter h and norms (or seminorms), but it has been very difficult to evaluate such constants explicitly. For quantitative purposes, however, it is indispensable to evaluate or bound them as accurately as possible, because sharper estimates enable more efficient finite element computations. Thus such evaluation has become progressively more important and has been attempted especially for adaptive finite element calculations based on a posteriori error estimation as well as for numerical verification by FEM [3, 6, 8, 10, 25]. In this paper, we will give some fundamental results on various interpolation error constants of the most popular triangular finite elements.

^{*}Corresponding author. Tel.: +81 3 5465 8345; fax: +81 3 5465 7011; e-mail: kikuchi@ms.u-tokyo.ac.jp

More specifically, we derive some fundamental estimates for the interpolation error constants appearing in the popular P_0 (piecewise constant) and P_1 (piecewise linear) triangular finite elements. Inspired by the monumental paper of Babuška-Aziz [5], we analyze the dependence of several constants on the geometric parameters such as the maximum interior angle and the minimum edge length of a triangle more quantitatively than works precedent to ours. Among them, the optimal constant (C_4 in the present paper) appearing in the H^1 error estimate of the P_1 interpolation of H^2 functions over the unit isosceles right triangle is essential and frequently used, and it was explicitly evaluated firstly by Natterer [27]. On the other hand, this constant was shown to be closely related to the one (C_1 in this paper) presented and effectively used by Babuška and Aziz in conjunction with the maximum angle condition [5]. More precisely, C_1 gives an upper bound quite close to the optimal constant C_4 , and the relation between C_4 and C_1 was further discussed in [25, 30]. Thus a precise estimation of these two constants is very important, and a number of researchers have given bounds for these using various approximation methods including numerical verification, see e.g. [4, 7, 22, 25, 26, 27, 30]. Furthermore, these constants can be also used to evaluate the interpolation error constants for the non-conforming P_1 triangle, as was mentioned in [19].

For the above Babuška-Aziz constant, we already succeeded in obtaining a value which is in a sense optimal [18]. That is, by analytically solving an eigenvalue problem for the 2D Laplacian over the above triangular domain, we showed that the constant can be easily determined from a solution of the simple transcendental equation $\mu^{-1} + \tan(\mu^{-1}) = 0$. In this process, we used the reflection (or symmetry) method [28]. In this paper, we will also give some additional results for exact values or bounds of various error constants. Moreover, we will present some explicit relations for the dependence of such constants on the geometry of triangles. In particular, emphasis is put on the maximum angle condition presented in [5]. We also give some analytical results based on asymptotic analysis with regards to the behaviors when a right triangle becomes very thin or slender. Such behaviors can be important for example in anisotropic triangulations, cf. [8].

Thus our results can be effectively used in the quantitative a priori and a posteriori error estimations of the finite element solutions by the P_1 triangular element and also those based on the P_0 triangle. The former is of course the most classical and fundamental one, but still in frequent use, while the latter appears in some mixed finite element methods and implicitly on various occasions. Moreover, we also give some concrete a posteriori error estimates to eigenvalues related to several error constants. Numerical results are also obtained for the error constants and a posteriori estimates of some eigenvalues.

The plan of this paper is as follows. Section 1 is the present one on some historical remarks and overview of our analysis. Section 2 gives necessary notations and concepts, and also introduces various error constants to be analyzed. Section 3 deals with estimation of various interpolation error constants, and Section 4 analyzes asymptotic behaviors of the error constants when the triangle is a thin right one. Section 5 gives application of our results to a posteriori estimation of some error constants by using the P_1 FEM. Section 6 is the one for numerical results, and Section 7 gives some concluding remarks and acknowledgements. Appendix is also attached to give some additional theoretical and numerical results related to Section 4.

2 Preliminaries: error constants

Let h, α and θ be positive constants such that

$$h > 0, \ 0 < \alpha \le 1, \ (\frac{\pi}{3} \le) \cos^{-1} \frac{\alpha}{2} \le \theta < \pi.$$
 (1)

Then we can define the triangle $T_{\alpha,\theta,h}$ by $\triangle OAB$ with three vertices O(0,0), A(h,0) and $B(\alpha h \cos \theta, \alpha h \sin \theta)$. From (1), AB turns out to be the edge of maximum length, i. e. $\overline{AB} \ge h \ge \alpha h$ with $h = \overline{OA}$ and $\alpha h = \overline{OB}$ being the medium and the minimum edge lengths, respectively. It is to be noted here that the notation h is mostly used as the largest edge length in standard textbooks such as [11], but our usage of h as the medium one may be convenient for the present purposes. A point on the closure of $T_{\alpha,\theta,h}$ is denoted by $x = \{x_1, x_2\}$, and the three edges e_1, e_2 and e_3 of $T_{\alpha,\theta,h}$ are defined as

$$e_1 = OA, \ e_2 = OB, \ e_3 = AB.$$
 (2)

We can configure any triangle as $T_{\alpha,\theta,h}$ with some α, θ and h, by using an appropriate congruent transformation in \mathbb{R}^2 . As the usage in [5], we will use abbreviated notations $T_{\alpha,\theta} = T_{\alpha,\theta,1}$, $T_{\alpha} = T_{\alpha,\pi/2}$ and $T = T_1$ (Fig. 1).



Figure 1: Notations for triangles : $T_{\alpha,\theta} = T_{\alpha,\theta,1}$, $T_{\alpha} = T_{\alpha,\pi/2}$, $T = T_1$

We will use the popular Hilbert space $L_2(T_{\alpha,\theta,h})$, and denote its norm by $\|\cdot\|_{T_{\alpha,\theta,h}}$, where the subscript $T_{\alpha,\theta,h}$ is often omitted if there is no fear of confusion. When we need to use the L_2 space and its norm for other domains such as Ω , we will use notations like $L_2(\Omega)$ and $\|\cdot\|_{\Omega}$. Let us define the following closed linear spaces for functions over $T_{\alpha,\theta,h}$:

$$V_{\alpha,\theta,h}^{0} = \{ v \in H^{1}(T_{\alpha,\theta,h}) \mid \int_{T_{\alpha,\theta,h}} v(x) \, dx = 0 \},$$
(3)

$$V_{\alpha,\theta,h}^{i} = \{ v \in H^{1}(T_{\alpha,\theta,h}) \mid \int_{e_{i}} v \, ds = 0 \} \quad (i = 1, 2, 3),$$
(4)

$$V_{\alpha,\theta,h}^{4} = \{ v \in H^{2}(T_{\alpha,\theta,h}) | v(O) = v(A) = v(B) = 0 \},$$
(5)

where $H^1(T_{\alpha,\theta,h})$ and $H^2(T_{\alpha,\theta,h})$ are respectively the first- and second-order Sobolev spaces for real square integrable functions over $T_{\alpha,\theta,h}$ [2], and ds is the line element. For other domains like Ω , we will also use spaces such as $H^1(\Omega)$ and $H^2(\Omega)$ later. For the above spaces, we will again use abbreviated notations $V_{\alpha,\theta}^i = V_{\alpha,\theta,1}^i$, $V_{\alpha}^i = V_{\alpha,\pi/2}^i$ and $V^i = V_1^i$ $(0 \le i \le 4)$. Let us consider the usual P_0 interpolation operator $\Pi_{\alpha,\theta,h}^0$ and P_1 one $\Pi_{\alpha,\theta,h}^1$ for functions

on $T_{\alpha,\theta,h}$ [10, 11, 21]: $\Pi^0_{\alpha,\theta,h}v$ for $\forall v \in H^1(T_{\alpha,\theta,h})$ is a constant function well-defined by

$$(\Pi^{0}_{\alpha,\theta,h}v)(x) = \int_{T_{\alpha,\theta,h}} v(y) \, dy \left/ \int_{T_{\alpha,\theta,h}} dy \quad (\forall x \in T_{\alpha,\theta,h}) \right.$$
(6)

while $\Pi^1_{\alpha,\theta,h}v$ for $\forall v \in H^2(T_{\alpha,\theta,h})$ is an at most linear polynomial function such that

$$(\Pi^1_{\alpha,\theta,h}v)(x) = v(x) \text{ for } x = O, A, B.$$
(7)

To give error estimates for these interpolation operators, it is natural to evaluate the positive constants defined by

$$C_i(\alpha, \theta, h) = \sup_{v \in V_{\alpha, \theta, h}^i \setminus \{0\}} \frac{\|v\|}{|v|_1} \quad (i = 0, 1, 2, 3),$$
(8)

$$C_4(\alpha, \theta, h) = \sup_{v \in V^4_{\alpha, \theta, h} \setminus \{0\}} \frac{|v|_1}{|v|_2},$$
(9)

$$C_5(\alpha, \theta, h) = \sup_{v \in V_{\alpha, \theta, h}^4 \setminus \{0\}} \frac{\|v\|}{|v|_2},$$
(10)

where $|v|_1 = (\sum_{i=1}^2 \|\partial v / \partial x_i\|^2)^{1/2}$, and $|v|_2 = (\sum_{i,j=1}^2 \|\partial^2 v / \partial x_i \partial x_j\|^2)^{1/2}$. When we need to specify a domain like Ω for the above semi-norms $|\cdot|_1$ and $|\cdot|_2$, we will use $|\cdot|_{1,\Omega}$ and $|\cdot|_{2,\Omega}$, respectively. The existence of these positive constants follows from the Rellich compactness theorem. Due to the properties to become clear soon, such constants together with some related ones are often called *interpolation error constants*. We will again use abbreviated notations $C_i(\alpha, \theta) = C_i(\alpha, \theta, 1), C_i(\alpha) = C_i(\alpha, \pi/2) \text{ and } C_i = C_i(1) \text{ for } 0 \le i \le 5.$

By a simple scale change, we find that $C_i(\alpha, \theta, h) = hC_i(\alpha, \theta)$ (i = 0, 1, 2, 3, 4) and $C_5(\alpha, \theta, h) = h^2 C_5(\alpha, \theta)$. These relations and constants are used to derive popular interpolation error estimates for $\Pi^{i}_{\alpha,\theta,h}$ (i = 0, 1) applied to functions on $T_{\alpha,\theta,h}$ [10, 11, 21]:

$$\|v - \Pi^0_{\alpha,\theta,h}v\| \le C_0(\alpha,\theta)h|v|_1 \qquad ; \forall v \in H^1(T_{\alpha,\theta,h}),$$
(11)

$$|v - \Pi^1_{\alpha,\theta,h}v|_1 \le C_4(\alpha,\theta)h|v|_2 \qquad ; \forall v \in H^2(T_{\alpha,\theta,h}),$$
(12)

$$\|v - \Pi^1_{\alpha,\theta,h}v\| \le C_5(\alpha,\theta)h^2|v|_2 \quad ; \forall v \in H^2(T_{\alpha,\theta,h}),$$
(13)

where we have used the facts that $v - \Pi^0_{\alpha,\theta,h} v \in V^0_{\alpha,\theta,h}$ for $v \in H^1(T_{\alpha,\theta,h})$ and $v - \Pi^1_{\alpha,\theta,h} v \in V^4_{\alpha,\theta,h}$ for $v \in H^2(T_{\alpha,\theta,h})$.

Moreover, for the partial derivative $\partial v / \partial x_1$ of $v \in H^2(T_{\alpha,\theta,h})$, we have

$$\left\|\frac{\partial(v - \Pi^{1}_{\alpha,\theta,h}v)}{\partial x_{1}}\right\| \leq C_{1}(\alpha,\theta)h\left|\frac{\partial v}{\partial x_{1}}\right|_{1}, \qquad (14)$$

since $\partial(v - \prod_{\alpha,\pi/2,h}^{1} v) / \partial x_1 \in V_{\alpha,\theta,h}^1$. On the other hand, to obtain an estimate in terms of $C_2(\alpha, \theta)$, we introduce rotation of the x_1 - x_2 plane around the origin O by angle $\theta - \pi/2$ so that the edge OB becomes the ordinate. Then the coordinate transformation $\hat{x} = \Phi_{\theta}(x)$ between the original variable $x = \{x_1, x_2\}$ and the new one $\hat{x} = \{\hat{x}_1, \hat{x}_2\}$ is given by, together with the associated transformation $\hat{v} = v \circ \Phi_{\theta}^{-1}$ for $v \in H^2(T_{\alpha,\theta,h})$,

$$\hat{x}_1 = x_1 \sin \theta - x_2 \cos \theta, \quad \hat{x}_2 = x_1 \cos \theta + x_2 \sin \theta, \quad (15)$$

$$\hat{v}(\hat{x}) = v(x) = v(\hat{x}_1 \sin \theta + \hat{x}_2 \cos \theta, -\hat{x}_1 \cos \theta + \hat{x}_2 \sin \theta).$$
(16)

Based on essentially the same arguments as for $\partial v / \partial x_1$, we can show for $\partial \hat{v} / \partial \hat{x}_2$ that

$$\left\|\frac{\partial(\hat{v} - \hat{\Pi}^{1}_{\alpha,\theta,h}\hat{v})}{\partial\hat{x}_{2}}\right\| \leq C_{2}(\alpha,\theta)h\left|\frac{\partial\hat{v}}{\partial\hat{x}_{2}}\right|_{1}, \qquad (17)$$

where $\hat{\Pi}^1_{\alpha,\theta,h}$ is $\Pi^1_{\alpha,\theta,h}$ for the rotated $T_{\alpha,\theta,h}$. The above two estimates (14) and (17) are in a sense sharper than (12) as noted in [21]. Similar relation also holds for $C_3(\alpha,\theta)$.

Remark 1. Refining the above arguments, we can obtain various anisotropic error estimates [8] such as

$$|v - \Pi^{1}_{\alpha,\theta,h}v|_{1} \le h \sqrt{\sum_{i,j=1}^{2} c_{ij} \left\| \frac{\partial^{2} v}{\partial x_{i} x_{j}} \right\|^{2}},$$

where c_{ij} 's $(1 \le i, j \le 2)$ are constants similar to $C_k(\alpha, \theta)$'s $(0 \le k \le 5)$ and can take unequal values. We do not make such a refinement here, though it can be an interesting subject.

Thus we can give quantitative interpolation estimates, provided that we succeed in evaluating or bounding the constants $C_i(\alpha, \theta)$'s explicitly. So we will try to bound these constants by fairly simple functions of α and θ . Notice here that each of such constants can be characterized by minimization of a kind of Rayleigh quotient. Then it is equivalent to finding the minimum eigenvalue of a certain eigenvalue problem expressed by a weak formulation, which is further expressed by a partial differential equation with some auxiliary conditions. More specifically, we can characterize the constants $C_i(\alpha, \theta)$'s by minimization of Rayleigh's quotients $R_{\alpha,\theta}^{(i)}$'s :

$$C_i^{-2}(\alpha, \theta) = \inf_{v \in V_{\alpha, \theta}^i \setminus \{0\}} R_{\alpha, \theta}^{(i)}(v) \; ; \; \; R_{\alpha, \theta}^{(i)}(v) = \frac{\|v\|_1^2}{\|v\|^2} \qquad (i = 0, 1, 2, 3) \; , \tag{18}$$

$$C_4^{-2}(\alpha, \theta) = \inf_{v \in V_{\alpha, \theta}^4 \setminus \{0\}} R_{\alpha, \theta}^{(4)}(v) \; ; \; \; R_{\alpha, \theta}^{(4)}(v) = \frac{|v|_2^2}{|v|_1^2} \,, \tag{19}$$

$$C_5^{-2}(\alpha,\theta) = \inf_{v \in V_{\alpha,\theta}^4 \setminus \{0\}} R_{\alpha,\theta}^{(5)}(v) \; ; \; \; R_{\alpha,\theta}^{(5)}(v) = \frac{|v|_2^2}{\|v\|^2} \,, \tag{20}$$

where all notations and functions are for $T_{\alpha,\theta}$.

By the standard compactness arguments, each infimum above is actually a minimum, and is the smallest eigenvalue of a certain eigenvalue problem. For example, the eigenvalue problem associated with $C_0(\alpha, \theta)$ is to find $\lambda \in \mathbf{R}$ and $u \in V^0_{\alpha, \theta} \setminus \{0\}$ that satisfy

$$(\nabla u, \nabla v)_{T_{\alpha,\theta}} = \lambda(u, v)_{T_{\alpha,\theta}} \quad (\forall v \in V^0_{\alpha,\theta}).$$
(21)

Here, $(\cdot, \cdot)_{T_{\alpha,\theta}}$ denotes the inner products of both $L_2(T_{\alpha,\theta})$ and $L_2(T_{\alpha,\theta})^2$, and ∇ is the gradient operator. When we consider the corresponding inner products for domains like Ω , we will use notations such as $(\cdot, \cdot)_{\Omega}$. The present eigenvalue problem is also expressed by a partial differential equation, a linear constraint for $V_{\alpha,\theta}^0$ and a boundary condition [25, 26]:

$$-\Delta u = \lambda u \text{ in } T_{\alpha,\theta}, \quad \int_{T_{\alpha,\theta}} u(x) \, dx = 0, \quad \frac{\partial u}{\partial n} = 0 \text{ on } \partial T_{\alpha,\theta}, \tag{22}$$

where $\frac{\partial}{\partial n}$ denotes the outward normal derivative on edges, and $\partial T_{\alpha,\theta}$ does the boundary of $T_{\alpha,\theta}$. The above boundary condition is the homogeneous Neumann one, and the desired minimum eigenvalue is the second (and positive) one for the same problem without the linear constraint.

For $C_1(\alpha, \theta)$, it is characterized in essentially the same fashion as (21), if the associated space $V^0_{\alpha,\theta}$ is replaced with $V^1_{\alpha,\theta}$. On the other hand, the equations corresponding to (22) become more complicated [25, 26]:

$$-\Delta u = \lambda u \text{ in } T_{\alpha,\theta}, \quad \int_0^1 u(x_1,0) \, dx_1 = 0, \quad \frac{\partial u}{\partial n} = \begin{cases} 0 & \text{on edges } OB \text{ and } AB, \\ c & \text{on edge } OA, \end{cases}$$
(23)

where c denotes an unknown constant to be decided simultaneously with u and λ . See also Section 5.3 of this paper.

The other constants are characterized similarly. For example, the eigenvalue problem associated to $C_4(\alpha, \theta)$ is to find $\lambda \in \mathbf{R}$ and $u \in V^4_{\alpha, \theta} \setminus \{0\}$ that satisfy

$$\sum_{i,j=1}^{2} (\partial^2 u / \partial x_i \partial x_j, \partial^2 v / \partial x_i \partial x_j)_{T_{\alpha,\theta}} = \lambda (\nabla u, \nabla v)_{T_{\alpha,\theta}} \quad (\forall v \in V_{\alpha,\theta}^4) .$$
(24)

But the partial differential equation related to the above and also that to $C_5(\alpha, \theta)$ are of fourth order ones with special linear constraints and boundary conditions, and are more difficult to deal with than the second order equations as in (22) and (23), cf. [4, 7]. Since $T_{\alpha,\theta}$ is a triangle, it is difficult to solve such eigenvalue problems explicitly even in the case of second-order equations, except in some rare cases to be shown later.

3 Estimation of interpolation error constants

It is in general difficult to obtain exact values of the error constants $C_i(\alpha, \theta)$'s except for very rare cases. In this section, we will first give some formulas to bound them in terms of their special values such as $C_i = C_i(1, \frac{\pi}{2}, 1)$'s. Such formula can be useful for various purposes, provided that the selected special values are evaluated with sufficient accuracy. So, we will also perform exact evaluation of some special constants.

3.1 Reconsideration of Natterer's results

Natterer [27] derived an upper bound formula for $C_4(\alpha, \theta)$ in terms of $C_4 = C_4(1, \frac{\pi}{2}, 1)$, α and θ . He also gave an upper bound for C_4 , so that his formula has been effectively used in quantitative error estimates of finite element solutions including numerical verifications of various differential equations [25, 26]. Here we begin by applying his techniques to bound the error constants introduced in Section 2.

To this end, let us introduce the following simple affine transformation $\xi = \Psi_{\alpha,\theta}(x)$ between $x = \{x_1, x_2\} \in T_{\alpha,\theta}$ and $\xi = \{\xi_1, \xi_2\} \in T = T_{1,\frac{\pi}{2},1}$:

$$\xi_1 = x_1 - \frac{x_2 \cos \theta}{\sin \theta}, \ \xi_2 = \frac{x_2}{\alpha \sin \theta}; \ x_1 = \xi_1 + \alpha \xi_2 \cos \theta, \ x_2 = \alpha \xi_2 \sin \theta.$$
 (25)

By eigenvalue analysis of matrices resulting from the above transformation in the Rayleigh quotients from (18) through (20), we obtain the following results.

Theorem 1. For $\alpha \in]0, +\infty[$ and $\theta \in]0, \pi[$, $C_i(\alpha, \theta)$'s are bounded as

$$\psi_i(\alpha, \theta)C_i \le C_i(\alpha, \theta) \le \phi_i(\alpha, \theta)C_i \quad (0 \le i \le 5),$$
(26)

where $C_i = C_i(1, \frac{\pi}{2}) \ (0 \le i \le 5)$,

$$\psi_i(\alpha,\theta) = \sqrt{\frac{\nu_-(\alpha,\theta)}{2}} \quad (0 \le i \le 3), \ \psi_4(\alpha,\theta) = \frac{\nu_-(\alpha,\theta)}{\sqrt{2\nu_+(\alpha,\theta)}}, \ \psi_5(\alpha,\theta) = \frac{\nu_-(\alpha,\theta)}{2}, \ (27)$$

$$\phi_i(\alpha,\theta) = \sqrt{\frac{\nu_+(\alpha,\theta)}{2}} \quad (0 \le i \le 3), \ \phi_4(\alpha,\theta) = \frac{\nu_+(\alpha,\theta)}{\sqrt{2\nu_-(\alpha,\theta)}}, \ \phi_5(\alpha,\theta) = \frac{\nu_+(\alpha,\theta)}{2}$$
(28)

with

$$\nu_{-}(\alpha,\theta) = 1 + \alpha^{2} - \sqrt{1 + 2\alpha^{2}\cos 2\theta + \alpha^{4}}, \ \nu_{+}(\alpha,\theta) = 1 + \alpha^{2} + \sqrt{1 + 2\alpha^{2}\cos 2\theta + \alpha^{4}}.$$
 (29)

Remark 2. In general, the upper bounds are more important than the lower ones, but the latter may be meaningful in evaluating the accuracy or efficiency of the boundings. The above estimates should be used essentially in the range $0 < \alpha \leq 1$ and $\frac{\pi}{3} \leq \cos^{-1} \frac{\alpha}{2} \leq \theta < \pi$, although they may be meaningful for general triangles without restrictions in (1), and the upper bounds except for i = 4 are uniformly bounded there. This fact means that these constants are robust to deformation of the triangle $T_{\alpha,\theta}$. On the other hand, the upper bound for $C_4(\alpha, \theta)$ is not so, and hence, to assure such uniform boundedness, we need the so-called minimum

angle condition [11]: the minimum angle of $T_{\alpha,\theta}$ is bounded from below by a certain positive constant. This may be seen by using the identity $\nu_{-}(\alpha,\theta)\nu_{+}(\alpha,\theta) = 4\alpha^{2}\sin^{2}\theta$ and rewriting the upper bound inequality as

$$C_4(\alpha, \theta) \le \frac{C_4}{\alpha \sin \theta} \left(\frac{\nu_+(\alpha, \theta)}{2}\right)^{\frac{3}{2}} .$$
(30)

Namely, we can see that the right-hand side diverges to $+\infty$ as $\alpha \to +0$ for each fixed $\theta \in]0, \pi[$. On the other hand, the minimum edge length α of $T_{\alpha,\theta}$ cannot approach to 0 under the minimum angle condition. It is also to be noted that the above inequality for $C_4(\alpha, \theta)$ is exactly the same as obtained by Natterer [27], although our notations are different from his. Actually, $C_4(\alpha, \theta)$ is uniformly bounded under the maximum angle condition of Babuška-Aziz [5], which requires the maximum interior angle of $T_{\alpha,\theta}$ to be away from π by a positive constant and hence is weaker than the usual minimum angle one [11]. It is also known that this weaker condition is essential and cannot be relaxed any more [5].

Proof. We will use the coordinate transformation (25) between $T_{\alpha,\theta}$ and T. By simple calculations, we have for $\tilde{v}(\xi_1,\xi_2) = v(x_1,x_2)$, i. e., $\tilde{v} = v \circ \Psi_{\alpha,\theta}^{-1}$ under the present transformation :

$$\sum_{i=1}^{2} \left(\frac{\partial v}{\partial x_{i}}\right)^{2} = \frac{1}{\sin^{2}\theta} \left[\left(\frac{\partial \tilde{v}}{\partial \xi_{1}}\right)^{2} - \frac{2\cos\theta}{\alpha} \frac{\partial \tilde{v}}{\partial \xi_{1}} \frac{\partial \tilde{v}}{\partial \xi_{2}} + \frac{1}{\alpha^{2}} \left(\frac{\partial \tilde{v}}{\partial \xi_{2}}\right)^{2} \right] ,$$

where v and \tilde{v} are assumed to be sufficiently smooth. The two eigenvalues associated to the quadratic form for $\partial \tilde{v} / \partial \xi_i$ (i = 1, 2) in $[\cdot]$ above are two solutions of the characteristic equation $\mu^2 - \alpha^{-2}(1+\alpha^2)\mu + \alpha^{-2}\sin^2\theta = 0$, and are given by $\nu_-(\alpha, \theta)/(2\alpha^2)$ and $\nu_+(\alpha, \theta)/(2\alpha^2)$. Thus we can easily derive

$$\frac{\nu_{-}(\alpha,\theta)}{2\alpha^{2}\sin^{2}\theta}\sum_{i=1}^{2}\left(\frac{\partial\tilde{v}}{\partial\xi_{i}}\right)^{2} \leq \sum_{i=1}^{2}\left(\frac{\partial v}{\partial x_{i}}\right)^{2} \leq \frac{\nu_{+}(\alpha,\theta)}{2\alpha^{2}\sin^{2}\theta}\sum_{i=1}^{2}\left(\frac{\partial\tilde{v}}{\partial\xi_{i}}\right)^{2} \,.$$

Moreover, the Jacobian of the present transformation is evaluated as $\partial(x_1, x_2)/\partial(\xi_1, \xi_2) = \alpha \sin \theta$. From these estimates and the identity $\nu_-(\alpha, \theta)\nu_+(\alpha, \theta) = 4\alpha^2 \sin^2 \theta$, we have

$$\|v\|_{T_{\alpha,\theta}}^{2} = \alpha \sin \theta \|\tilde{v}\|_{T}^{2}, \quad \frac{2\alpha \sin \theta}{\nu_{+}(\alpha,\theta)} |\tilde{v}|_{1,T}^{2} \le |v|_{1,T_{\alpha,\theta}}^{2} \le \frac{2\alpha \sin \theta}{\nu_{-}(\alpha,\theta)} |\tilde{v}|_{1,T}^{2}, \quad (a.1)$$

where $|\cdot|_{1,T_{\alpha,\theta}}$, for example, denotes $|\cdot|_1$ for $T_{\alpha,\theta}$. The results for i = 0, 1, 2, 3 are now easy to obtain by using the above and the definitions of the constants $C_i(\alpha, \theta)$'s, since the present transformation yields a bijection between $V^i_{\alpha,\theta}$ and V^i .

Similarly, we obtain

$$\sum_{i,j=1}^{2} \left(\frac{\partial^2 v}{\partial x_i \partial x_j} \right)^2 = \frac{1}{\sin^4 \theta} \left[\left(\frac{\partial^2 \tilde{v}}{\partial \xi_1^2} \right)^2 + \frac{1}{\alpha^4} \left(\frac{\partial^2 \tilde{v}}{\partial \xi_2^2} \right)^2 + \frac{2(1 + \cos^2 \theta)}{\alpha^2} \left(\frac{\partial^2 \tilde{v}}{\partial \xi_1 \partial \xi_2} \right)^2 \right. \\ \left. + \frac{2\cos^2 \theta}{\alpha^2} \frac{\partial^2 \tilde{v}}{\partial \xi_1^2} \frac{\partial^2 \tilde{v}}{\partial \xi_2^2} - \frac{4\cos \theta}{\alpha} \frac{\partial^2 \tilde{v}}{\partial \xi_1^2} \frac{\partial^2 \tilde{v}}{\partial \xi_1 \partial \xi_2} - \frac{4\cos \theta}{\alpha^3} \frac{\partial^2 \tilde{v}}{\partial \xi_2^2} \frac{\partial^2 \tilde{v}}{\partial \xi_1 \partial \xi_2} \right]$$

Let us consider the following real symmetric matrix related to the quadratic form for $\partial^2 \hat{v} / \partial \xi_1^2$, $\partial^2 \hat{v} / \partial \xi_2^2$ and $\sqrt{2} \partial^2 \hat{v} / \partial \xi_1 \partial \xi_2$ in $[\cdot]$ of the right-hand side above :

$$\begin{pmatrix} 1 & \frac{\cos^2\theta}{\alpha^2} & -\frac{\sqrt{2}\cos\theta}{\alpha} \\ \frac{\cos^2\theta}{\alpha^2} & \frac{1}{\alpha^4} & -\frac{\sqrt{2}\cos\theta}{\alpha^3} \\ -\frac{\sqrt{2}\cos\theta}{\alpha} & -\frac{\sqrt{2}\cos\theta}{\alpha^3} & \frac{1+\cos^2\theta}{\alpha^2} \end{pmatrix}$$

The associated characteristic equation is

$$\mu^{3} - \alpha^{-4} \{ 1 + (1 + \cos^{2}\theta)\alpha^{2} + \alpha^{4} \} \mu^{2} + \alpha^{-6} \sin^{2}\theta \{ 1 + (1 + \cos^{2}\theta)\alpha^{2} + \alpha^{4} \} \mu - \alpha^{-6} \sin^{6}\theta \\ = (\mu - \alpha^{-2} \sin^{2}\theta) \{ \mu^{2} - \alpha^{-4} (1 + 2\alpha^{2} \cos^{2}\theta + \alpha^{4})\mu + \alpha^{-4} \sin^{4}\theta \} = 0 ,$$

which has three eigenvalues $\nu_{-}^{2}(\alpha,\theta)/(4\alpha^{4}) \leq \alpha^{-2} \sin^{2}\theta \leq \nu_{+}^{2}(\alpha,\theta)/(4\alpha^{4})$ with $\nu_{-}(\alpha,\theta)$ and $\nu_{+}(\alpha,\theta)$ defined by (29). Now we have the estimates

$$\frac{\nu_{-}^{2}(\alpha,\theta)}{4\alpha^{4}\sin^{4}\theta}\sum_{i,j=1}^{2}\left(\frac{\partial^{2}\tilde{v}}{\partial\xi_{i}\partial\xi_{j}}\right)^{2} \leq \sum_{i,j=1}^{2}\left(\frac{\partial^{2}v}{\partial x_{i}\partial x_{j}}\right)^{2} \leq \frac{\nu_{+}^{2}(\alpha,\theta)}{4\alpha^{4}\sin^{4}\theta}\sum_{i,j=1}^{2}\left(\frac{\partial^{2}\tilde{v}}{\partial\xi_{i}\partial\xi_{j}}\right)^{2},$$

which gives, as (a.1),

$$\frac{4\alpha\sin\theta}{\nu_+^2(\alpha,\theta)}|\tilde{v}|_{2,T}^2 \le |v|_{2,T_{\alpha,\theta}}^2 \le \frac{4\alpha\sin\theta}{\nu_-^2(\alpha,\theta)}|\tilde{v}|_{2,T}^2 . \tag{a.2}$$

From (a.1) and (a.2), we obtain the results for i = 4, 5.

As a corollary of the preceding theorem, we can bound each $C_i(\alpha, \theta)$ in terms of $C_i(\alpha)$ and θ . Such estimates can be effective when the dependence of the considered $C_i(\alpha)$ on α is known as we will see later. The bounding can be achieved by introducing the following affine transformation between $x = \{x_1, x_2\} \in T_{\alpha, \theta}$ and $\xi = \{\xi_1, \xi_2\} \in T_{\alpha}$:

$$\xi_1 = x_1 - x_2 \cos \theta / \sin \theta , \quad \xi_2 = x_2 / \sin \theta , \tag{31}$$

which is a bit different from that used by Babuška and Aziz [5]. But essentially the same can be attained by simply applying the results of Theorem 1 for $\alpha = 1$, as may be seen by comparing (31) with (25).

Corollary 1. For $\alpha \in]0, +\infty[$ and $\theta \in]0, \pi[$, $C_i(\alpha, \theta)$'s are bounded as

$$\psi_i(\theta)C_i(\alpha) \le C_i(\alpha, \theta) \le \phi_i(\theta)C_i(\alpha) \quad (0 \le i \le 5),$$
(32)

where $\psi(\theta) = \psi(1, \theta)$ and $\phi(\theta) = \phi(1, \theta)$ for $0 \le i \le 5$. More specifically,

$$\psi_i(\theta) = \sqrt{1 - |\cos\theta|} \quad (0 \le i \le 3), \ \psi_4(\theta) = \frac{1 - |\cos\theta|}{\sqrt{1 + |\cos\theta|}}, \ \psi_5(\theta) = 1 - |\cos\theta|, \ (33)$$

$$\phi_i(\theta) = \sqrt{1 + |\cos \theta|} \quad (0 \le i \le 3), \ \phi_4(\theta) = \frac{1 + |\cos \theta|}{\sqrt{1 - |\cos \theta|}}, \ \phi_5(\theta) = 1 + |\cos \theta|.$$
(34)

Remark 3. The function form of $\phi_4(\theta)$ associated to $C_4(\alpha, \theta)$ is consistent with the maximum angle condition in [5], since $\phi_4(\theta)$ is bounded on $[\frac{\pi}{3}, \pi - \delta]$ for each sufficiently small fixed $\delta > 0$. Thus, $C_4(\alpha, \theta)$ is uniformly bounded for $0 < \alpha \le 1$ and $\frac{\pi}{3} \le \theta < \pi - \delta$, if we can show that $C_4(\alpha)$ is uniformly bounded for such α . Notice here that $C_4(\alpha) \le C_1 = C_2$ for $\alpha \le 1$, as will be seen in the subsequent section.

3.2 Estimation of $C_4(\alpha, \theta)$ by $C_1(\alpha, \theta)$ and $C_2(\alpha, \theta)$

We can also give an upper bound for $C_4(\alpha, \theta)$ in terms of $C_1(\alpha, \theta)$ and $C_2(\alpha, \theta)$.

Theorem 2. For $\forall \alpha \in]0, +\infty[$ and $\forall \theta \in]0, \pi[$, $C_4(\alpha, \theta)$ is bounded as

$$C_4(\alpha, \theta) \le \frac{1}{\sqrt{2}\sin\theta}\nu(\alpha, \theta) \le C_1 \frac{1 + |\cos\theta|}{\sin\theta} \sqrt{\frac{\nu_+(\alpha, \theta)}{2}}, \qquad (35)$$

where $\nu_{+}(\alpha, \theta)$ is defined by (29), and $\nu(\alpha, \theta)$ by

$$\nu(\alpha,\theta) = \left[C_1^2(\alpha,\theta) + C_2^2(\alpha,\theta) + 2C_1(\alpha,\theta)C_2(\alpha,\theta)\cos^2\theta + (C_1(\alpha,\theta) + C_2(\alpha,\theta))\sqrt{C_1^2(\alpha,\theta) + C_2^2(\alpha,\theta) + 2C_1(\alpha,\theta)C_2(\alpha,\theta)\cos 2\theta} \right]^{1/2}.$$
 (36)

In particular, the maximum angle condition applies to the present estimate (35), cf. [5, 21].

Remark 4. It is also possible to bound $C_4(\alpha, \theta)$ by $C_1(\alpha, \theta)$ and $C_3(\alpha, \theta)$ in a similar manner, although we omit the explicit expressions. It may be meaningful to compare two estimates (30) and (35) for $C_4(\alpha, \theta)$:

$$C_4(\alpha, \theta) \le \frac{C_4}{\alpha \sin \theta} \left(\frac{\nu_+(\alpha, \theta)}{2}\right)^{\frac{3}{2}} =: \gamma_1(\alpha, \theta),$$

$$C_4(\alpha, \theta) \le C_1 \frac{1 + |\cos \theta|}{\sin \theta} \sqrt{\frac{\nu_+(\alpha, \theta)}{2}} =: \gamma_2(\alpha, \theta).$$

Noting the relations $2\alpha |\cos \theta| \le \sqrt{1 + 2\alpha^2 \cos 2\theta + \alpha^4} \le 1 + \alpha^2$ and $2\alpha \le 1 + \alpha^2$, we find

$$\frac{\alpha(1+|\cos\theta|)}{1+\alpha^2}\frac{C_1}{C_4} \le \frac{\gamma_2(\alpha,\theta)}{\gamma_1(\alpha,\theta)} = \frac{2\alpha(1+|\cos\theta|)}{1+\alpha^2+\sqrt{1+2\alpha^2\cos 2\theta+\alpha^4}}\frac{C_1}{C_4} \le \frac{C_1}{C_4}.$$

It is known that $C_4 \approx 0.489$ by numerical computations without verification [4, 22, 30]. On the other hand, it is theoretically shown that C_1 is an upper bound of C_4 , but is quite close to C_4 as shown by numerically verified bounding $0.492 < C_1 < 0.493$ [25, 26, 18, 19], cf. also Theorems 3 and 4 later. Thus the above estimate shows that (35) is practically better than (30) for almost all values of α and θ . As a practical upper bound for C_4 , Siganevich [30] recommended 0.5. Proof. From the definition, we have

$$C_4^2(\alpha, \theta) = \sup_{v \in V_{\alpha, \theta}^4 \setminus \{0\}} \frac{|v|_1^2}{|v|_2^2} = \sup_{v \in V_{\alpha, \theta}^4 \setminus \{0\}} \frac{\|\partial v/\partial x_1\|^2 + \|\partial v/\partial x_2\|^2}{|\partial v/\partial x_1|_1^2 + |\partial v/\partial x_2|_1^2}.$$
 (b.1)

Recall here the transformation rules (15) and (16). Then, for the present $v \in V_{\alpha,\theta}^4 \setminus \{0\}$ and the associated $\hat{v} = v \circ \Phi_{\theta}^{-1}$, we can show as (14) and (17) that

$$\|\partial v/\partial x_1\| \le C_1(\alpha,\theta) |\partial v/\partial x_1|_1, \quad \|\partial \hat{v}/\partial \hat{x}_2\| \le C_2(\alpha,\theta) |\partial \hat{v}/\partial \hat{x}_2|_1, \qquad (b.2)$$

where $\partial \hat{v} / \partial \hat{x}_2 = \cos \theta \, \partial v / \partial x_1 + \sin \theta \, \partial v / \partial x_2$ at $x = \{x_1, x_2\} \in T_{\alpha, \theta}$ and $\hat{x} = \{\hat{x}_1, \hat{x}_2\} = \Phi_{\theta}(x)$. Then $\partial v / \partial x_2 = (\partial \hat{v} / \partial \hat{x}_2 - \cos \theta \, \partial v / \partial x_1) / \sin \theta$ can be evaluated as

$$\|\partial v/\partial x_2\|^2 \le \frac{1}{\sin^2 \theta} \left[\|\partial \hat{v}/\partial \hat{x}_2\|^2 + 2|\cos \theta| \cdot \|\partial \hat{v}/\partial \hat{x}_2\| \cdot \|\partial v/\partial x_1\| + \cos^2 \theta \|\partial v/\partial x_1\|^2 \right].$$

By (b.2) and the present inequality, we can bound $\|\partial v/\partial x_1\|^2 + \|\partial v/\partial x_2\|^2$ from above as

$$\begin{aligned} \|\partial v/\partial x_1\|^2 + \|\partial v/\partial x_2\|^2 \\ &\leq \frac{1}{\sin^2 \theta} \left[\|\partial v/\partial x_1\|^2 + 2|\cos \theta| \cdot \|\partial v/\partial x_1\| \cdot \|\partial \hat{v}/\partial \hat{x}_2\| + \|\partial \hat{v}/\partial \hat{x}_2\|^2 \right] \\ &\leq \frac{1}{\sin^2 \theta} \left[C_1^2(\alpha, \theta) |\partial v/\partial x_1|_1^2 + 2 C_1(\alpha, \theta) C_2(\alpha, \theta) |\cos \theta| \cdot |\partial v/\partial x_1|_1 \cdot |\partial \hat{v}/\partial \hat{x}_2|_1 \\ &+ C_2^2(\alpha, \theta) |\partial \hat{v}/\partial \hat{x}_2|_1^2 \right] . \end{aligned}$$
(b.3)

To evaluate $|\partial \hat{v}/\partial \hat{x}_2|_1$ above, we again use the relation $\partial \hat{v}/\partial \hat{x}_2 = \cos \theta \, \partial v/\partial x_1 + \sin \theta \, \partial v/\partial x_2$. Then we find

$$|\partial \hat{v}/\partial \hat{x}_2|_1 \le |\cos \theta| \cdot |\partial v/\partial x_1|_1 + \sin \theta |\partial v/\partial x_2|_1.$$

Substituting the above into the right-hand side of (b.3), we obtain

$$\begin{aligned} \|\partial v/\partial x_1\|^2 + \|\partial v/\partial x_2\|^2 \\ &\leq \frac{1}{\sin^2\theta} \left[\left\{ C_1^2(\alpha,\theta) + 2C_1(\alpha,\theta)C_2(\alpha,\theta)\cos^2\theta + C_2^2(\alpha,\theta)\cos^2\theta \right\} |\partial v/\partial x_1|_1^2 \right. \\ &\quad + 2C_2(\alpha,\theta) \left\{ C_1(\alpha,\theta) + C_2(\alpha,\theta) \right\}\sin\theta |\cos\theta| \cdot |\partial v/\partial x_1|_1 \cdot |\partial v/\partial x_2|_1 \\ &\quad + C_2^2(\alpha,\theta)\sin^2\theta \left|\partial v/\partial x_2|_1^2 \right] . \end{aligned}$$

By eigenvalue analysis of the quadratic form above for $|\partial v/\partial x_1|_1$ and $|\partial v/\partial x_2|_1$, we have, by using $\nu(\alpha, \theta)$ in (36),

$$\|\partial v/\partial x_1\|^2 + \|\partial v/\partial x_2\|^2 \le \frac{\nu^2(\alpha,\theta)}{2\sin^2\theta} \left[|\partial v/\partial x_1|_1^2 + |\partial v/\partial x_2|_1^2 \right],$$

which gives the former part of (35) by (b.1).

To derive the latter part of (35), we should use $C_i(\alpha, \theta) \leq \phi_i(\alpha, \theta)C_i$ (i = 1, 2) in (26) and the identities $\phi_1(\alpha, \theta) = \phi_2(\alpha, \theta) = \sqrt{\frac{\nu_+(\alpha, \theta)}{2}}$ and $C_1 = C_2$.

3.3 Determination of some constants

Theorem 1 tells us that we can obtain upper bounds of the constants $C_i(\alpha, \theta)$ $(0 \le i \le 5)$, if correct values of $C_i = C_i(1, \frac{\pi}{2}, 1)$ are known. The upper bounds thus evaluated may be rough but anyway correct, so that they can be used for various theoretical purposes. According to some preceding works [18, 19, 25, 26], such exact evaluation is possible at least for C_0 and $C_1 = C_2$. We will quote the results below, together with an additional result for C_3 .

Theorem 3. It holds for $C_i = C_i(1, \frac{\pi}{2}, 1) \ (0 \le i \le 3)$ that

- 1) $i = 0: C_0 = \frac{1}{\pi}$,
- 2) $i = 1, 2: C_1 = C_2$, and is given as the maximum positive solution of the transcendental equation for μ :

$$\frac{1}{\mu} + \tan \frac{1}{\mu} = 0.$$
 (37)

The concrete value of C_1 can be obtained numerically with verification. For example, we have the estimation

$$0.49282 < C_1 < 0.49293. \tag{38}$$

3)
$$i = 3$$
: $C_3 = \frac{C_1}{\sqrt{2}}$, $0.34847 < C_3 < 0.34856$.

Remark 5. i) Numerical computation without verification gives $C_1 = 0.49291245\cdots$ and $C_3 = 0.34854173\cdots$. The present transcendental equation can be commonly seen in vibration analysis of strings with special boundary conditions [28]. The constant C_1 plays an important role in various situations and is called the Babuška-Aziz constant in [18, 19].

ii) At present, exact values of C_4 and C_5 are not known to the best of the authors' knowledge. Fortunately, $C_1 (= C_2)$ is a nice upper bound of C_4 as we will see in Sections 4.2 and 6.2. Numerically, $C_4 \approx 0.489$ as was reported in [4, 22, 30]. As for C_5 , estimate $C_5 < 0.361$ is a correct but probably rough one given in [14], while an exact lower bound estimation is $C_5 \ge [(15 + \sqrt{193})/1440]^{1/2} = 0.1416...$, which is derived by the Ritz-Galerkin method using $x_1 + x_2 - x_1^2 - x_2^2$ and $x_1 x_2$ as the basis of the trial space employed in [25]. Our own numerical computations suggest that $C_5 \le 0.168$.

Proof. Refer [19, 25, 26] for 1) and [18, 19] for 2), respectively. For 3), we can prove by using the results for C_1 and a kind of symmetry method. We will proceed in three steps.

1] Similarly to (21), the eigenvalue problem associated to C_3 is given by: Find $\{\lambda, u\} \in \mathbf{R} \times V^3 \setminus \{0\}$ such that

$$(\nabla u, \nabla v)_T = \lambda(u, v)_T \quad (\forall v \in V^3).$$
(c.1)

Here, T is the unit right isosceles triangle $T_{1,\frac{\pi}{2},1}$, $V^3 = V_{1,\frac{\pi}{2},1}^3$ is defined by (4), and the inner products are those for T. Notice that we are interested only in the minimum eigenvalue and the associated eigenfunctions.

Let us divide T into two congruent parts using the line $x_2 = x_1$, which is also the line of symmetry for T. Moreover, one of the congruent parts is denoted by \tilde{T} :

$$T = \{x = \{x_1, x_2\} \in T; x_1 > x_2\}.$$

The eigenfunction $u \neq 0$ can be uniquely decomposed into the symmetric part u_s and the antisymmetric one u_a :

$$u = u_s + u_a \,,$$

where the symmetry and antisymmetry are those with respect to $x_2 = x_1$. Due to the orthogonalities of u_s and u_a for the bilinear forms $(\cdot, \cdot)_T$ and $(\nabla \cdot, \nabla \cdot)_T$, u_s and u_a can be dealt with separately: u_s and u_a both belong to V^3 and satisfy (c.1) for the minimum eigenvalue λ .

2] We first consider the case where $u_s \neq 0$. We can see that the restriction \tilde{u} of u_s to \tilde{T} is not zero and satisfies the following eigenvalue problem related to \tilde{T} :

$$\tilde{u} \in \tilde{V}^3 \setminus \{0\}; \quad (\nabla \tilde{u}, \nabla \tilde{v})_{\tilde{T}} = \lambda(\tilde{u}, \tilde{v})_{\tilde{T}} \quad (\forall \tilde{v} \in \tilde{V}^3), \qquad (c.2)$$

where λ is identical to the former one, the inner products are the L_2 ones for \tilde{T} , and \tilde{V}^3 is defined by

$$\tilde{V}^3 = \{ \tilde{v} \in H^1(\tilde{T}); \int_0^{\frac{1}{2}} \tilde{v}(1-s,s) \, ds = 0 \}$$

Now we can see that this is essentially the same problem as the eigenvalue problem for $C_1(1, \frac{\pi}{2}, \frac{1}{\sqrt{2}})$, since \tilde{T} is congruent to $T_{1, \frac{\pi}{2}, \frac{1}{\sqrt{2}}}$. It is also fairly easy to see that the eigenpair for the minimum eigenvalue of (c.2) satisfies (c.1), if the eigenfunction is extended to whole T symmetrically with respect to $x_2 = x_1$. Thus \tilde{u} is an eigenfunction for the minimum eigenvalue of (c.2) in the present case. Then we find that $C_3 = C_1/\sqrt{2}$, since $C_1(\alpha, \theta, 1/\sqrt{2}) = C_1(\alpha, \theta)/\sqrt{2}$ as we have seen in Section 2. Of course, this conclusion is derived under the assumption that $u_s \neq 0$.

3] Secondly, we consider the case where $u_a \neq 0$. Due to the antisymmetry, the trace of u_a to the line of symmetry $x_2 = x_1$ inside T is shown to be 0. Moreover, any antisymmetric function in $H^1(T)$ automatically satisfies the line integration condition imposed on V^3 . Thus the restriction u^{\dagger} of u_a to \tilde{T} is not zero and is an eigenfunction of the eigenvalue problem:

$$u^{\dagger} \in V^{\dagger} \setminus \{0\}; \quad (\nabla u^{\dagger}, \nabla v^{\dagger})_{\tilde{T}} = \lambda (u^{\dagger}, v^{\dagger})_{\tilde{T}} \quad (\forall v^{\dagger} \in V^{\dagger}), \qquad (c.3)$$

where λ is identical to the former one, and V^{\dagger} is defined by

$$V^{\dagger} = \{ v^{\dagger} \in H^{1}(\tilde{T}) ; v^{\dagger}(s,s) = 0 \ (0 < s < \frac{1}{2}) \}.$$

If we consider the reflection with respect to the line $x_1 = 1/2$, (c.3) becomes the problem of the same form with V^{\dagger} replaced by

$$V^* = \{ v^* \in H^1(\tilde{T}) ; v^*(1-s,s) = 0 \ (0 < s < \frac{1}{2}) \}.$$

Clearly, the eigenvalues remain the same under such a transformation. Since $V^* \subset \tilde{V}^3$, the minimum eigenvalue of (c.3) cannot be smaller than that of (c.2), as can be seen by considering the characterization of the minimum eigenvalue by the Rayleigh quotient. Thus it is sufficient to consider the case where $u_s \neq 0$ only, and the proof is complete.

3.4 Application to interpolation and a priori error estimates

In this subsection, we show how to apply the obtained results to interpolation error estimates and some a priori error estimates for FEM.

From the preceding considerations, especially equations (11) through (13) and Theorems 1 and 2, we have for example the following P_0 and P_1 interpolation error estimates :

$$\|v - \Pi^0_{\alpha,\theta,h}v\| \le C_0\phi_0(\alpha,\theta)h|v|_1 \quad ; \forall v \in H^1(T_{\alpha,\theta,h}),$$
(39)

$$|v - \Pi^1_{\alpha,\theta,h}v|_1 \le C_1 \frac{1 + |\cos\theta|}{\sin\theta} \left[\frac{\nu_+(\alpha,\theta)}{2}\right]^{\frac{1}{2}} h|v|_2 \quad ; \forall v \in H^2(T_{\alpha,\theta,h}),$$
(40)

$$\|v - \Pi^1_{\alpha,\theta,h}v\| \le C_5\phi_5(\alpha,\theta)h^2|v|_2 \quad ; \forall v \in H^2(T_{\alpha,\theta,h}).$$

$$\tag{41}$$

These may be rough but are still correct quantitative upper bounds, provided that the values of C_0 , C_1 and C_5 or at least their upper bounds are known. For C_0 and C_1 , we have obtained exact values in Theorem 3, while, presumably, C_5 has been evaluated only approximately as was noted in Remark 5.

As was already noted, such error bounds are available for triangles of general configuration by applying appropriate congruent transformations [5, 10, 11, 21]. Then such interpolation error estimates can be directly used in a priori error estimates of finite element solutions. In what follows, we will briefly explain an example of such process. See e. g. [11] for the details.

As a model problem, let us consider the Dirichlet problem of the Poisson equation over an bounded polygonal domain $\Omega \subset \mathbf{R}^2$: given $f \in L_2(\Omega)$, find $u \in H_0^1(\Omega)$ such that $-\Delta u = f$ in Ω . Here, $H_0^1(\Omega)$ is the popular subspace of $H^1(\Omega)$ with the homogeneous Dirichlet condition imposed. In the standard weak formulation, the condition for $u \in H_0^1(\Omega)$ is stated as

$$(\nabla u, \nabla v)_{\Omega} = (f, v)_{\Omega} \quad \left(\forall v \in H_0^1(\Omega)\right) , \tag{42}$$

where the L_2 -type inner products are those for Ω . As is well known, this is a well-posed problem so that we can define an operator G by $G : f \in L_2(\Omega) \mapsto u \in H_0^1(\Omega)$.

To apply the FEM based on the P_1 triangle to this problem, we consider a regular family of triangulations $\{\mathcal{T}^{\eta}\}_{\eta>0}$ of Ω , and then construct a P_1 finite element space $V^{\eta} \subset H_0^1(\Omega)$ for each \mathcal{T}^{η} , cf. [11] for the terminology *regular*. Here we have used η instead of the standard notation h for the discretization parameter, which has been already used in a different meaning. The finite element approximation $u_{\eta} \in V^{\eta}$ of the above u is now uniquely determined by imitating (42) in V^{η} :

$$(\nabla u_{\eta}, \nabla v_{\eta})_{\Omega} = (f, v_{\eta})_{\Omega} \quad (\forall v_{\eta} \in V^{\eta}).$$
(43)

An important fact in the error analysis of the Ritz-Galerkin FEM is the following best approximation property [11]:

$$|u - u_{\eta}|_{1,\Omega} = \min_{v_{\eta} \in V^{\eta}} |u - v_{\eta}|_{1,\Omega} , \qquad (44)$$

where $|\cdot|_{1,\Omega}$ is $|\cdot|_1$ for Ω (similar usages will frequently appear hereafter). Another important one is the L_2 -error estimate based on the Aubin-Nitsche trick [11]:

$$\|u - u_{\eta}\|_{\Omega} \le |u - u_{\eta}|_{1,\Omega} \inf_{v_{\eta} \in V^{\eta}} \sup_{g \in L_{2}(\Omega) \setminus \{0\}} \frac{|Gg - v_{\eta}|_{1,\Omega}}{\|g\|_{\Omega}}.$$
(45)

From (44), an error estimation based on the interpolation function $\Pi^{\eta,1}u \in V^{\eta}$ using the vertex values of u is given by

$$|u - u_{\eta}|_{1,\Omega} \le |u - \Pi^{\eta,1} u|_{1,\Omega} \,. \tag{46}$$

Clearly, the global interpolation operator $\Pi^{\eta,1}$ is closely related to the local one $\Pi^1_{\alpha,\theta,h}$. That is, for each triangle $K \in \mathcal{T}^{\eta}$, we can find a $T_{\alpha,\theta,h}$ congruent to K under a congruent transformation $\Phi_K : K \to T_{\alpha,\theta,h}$, and it then holds that $(\Pi^{\eta,1}u)|_K = [\Pi^1_{\alpha,\theta,h}\{(u|K) \circ \Phi_K^{-1}\}] \circ \Phi_K$. If $u \in H^2(\Omega)$, we have, using notations $\{\alpha_K, \theta_K, h_K\}$ for $\{\alpha, \theta, h\}$ of $T_{\alpha,\theta,h}$ associated to K,

$$|u - \Pi^{\eta,1}u|_{1,\Omega}^2 = \sum_{K \in \mathcal{T}^{\eta}} |u - \Pi^{\eta,1}u|_{1,K}^2 \le \sum_{K \in \mathcal{T}^{\eta}} h_K^2 C_4^2(\alpha_K, \theta_K) |u|_{2,K}^2.$$
(47)

Thus we obtain from (46) an a priori error estimate

$$|u - u_{\eta}|_{1,\Omega} \le |u - \Pi^{\eta,1} u|_{1,\Omega} \le C_{4,\eta} \eta |u|_{2,\Omega},$$
(48)

where $C_{4,\eta}$ and η are defined by

$$C_{4,\eta} = \max_{K \in \mathcal{T}^{\eta}} C_4(\alpha_K, \theta_K), \quad \eta = \max_{K \in \mathcal{T}^{\eta}} h_K.$$
(49)

To evaluate $C_{4,\eta}$ from above, we can utilize various upper bounds already derived for $C_4(\alpha, \theta)$, an example of which can be also found in (40). In problems more general than (42), we may also need upper bounds for $C_5(\alpha, \theta)$ to obtain global L_2 error bounds, although we can avoid the use of such bounds to a certain extent by adopting the Aubin-Nitsche trick [11]. The constants $C_i(\alpha, \theta)$ for $0 \le i \le 3$ may appear to be subsidiary here, but they actually play essential roles in the analysis of the non-conforming P_1 FEM as is noted in [19].

In order to apply the above to verification of various differential equations by FEM, it is often required to evaluate norms or semi-norms of the solutions by various data. A typical example is to give upper bounds of $|u|_{2,\Omega}$ in (48) by a norm of f. In the present case, we can use the well-known relation $|u|_{2,\Omega} \leq ||f||_{\Omega}$, provided that Ω is convex in addition to the assumptions already stated [15]. Then we have

$$|u - u_{\eta}|_{1,\Omega} \le C_{4,\eta} \eta ||f||_{\Omega} , \qquad (50)$$

and moreover, by applying (45) with v_{η} taken as $\Pi^{\eta,1}(Gg)$,

$$\|u - u_{\eta}\|_{\Omega} \le C_{4,\eta}\eta |u - u_{\eta}|_{1,\Omega} \le C_{4,\eta}^2 \eta^2 \|f\|_{\Omega},$$
(51)

where we have used the estimate $|Gg - \Pi^{\eta,1}(Gg)|_{1,\Omega} \leq C_{4,\eta}\eta |Gg|_{2,\Omega} \leq C_{4,\eta}\eta ||g||_{\Omega}$. The present estimation can be compared with the L_2 interpolation estimate

$$\|u - \Pi^{\eta,1}u\|_{\Omega} \le C_{5,\eta}\eta^2 \|u\|_{2,\Omega} \le C_{5,\eta}\eta^2 \|f\|_{\Omega} \quad \text{with} \ C_{5,\eta} = \max_{K \in \mathcal{T}^{\eta}} C_5(\alpha_K, \theta_K) \,.$$
(52)

Such evaluations become much more difficult for general problems, but have been gradually realized in various cases.

3.5 Application to a posteriori error estimates

A posteriori error estimation is also feasible and effective in various situations by using the interpolation error constants considered in the preceding subsections. So, before closing the present section, we also show how to apply the obtained results to a posteriori error estimates for FEM. Here we only explain a special and rather classical approach [12, 20, 24] briefly, but we can find a number of literatures on this subject.

Let q be an element of $H(\operatorname{div}; \Omega) := \{q \in L_2(\Omega)^2 \mid \operatorname{div} q \in L_2(\Omega)\}$ [12, 20]. Such q can be also chosen from the narrower space $H^1(\Omega)^2$. Then, some simple calculations give, with the same notations as those in Section 3.4,

$$\begin{aligned} |u - u_{\eta}|_{1,\Omega}^{2} &= (\nabla(u - u_{\eta}), \nabla(u - u_{\eta}))_{\Omega} = (u - u_{\eta}, -\Delta u)_{\Omega} - ((\nabla(u - u_{\eta}), \nabla u_{\eta})_{\Omega} \\ &= (u - u_{\eta}, f)_{\Omega} + ((\nabla(u - u_{\eta}), q - \nabla u_{\eta} - q)_{\Omega} \\ &= (u - u_{\eta}, f + \operatorname{div} q)_{\Omega} + ((\nabla(u - u_{\eta}), q - \nabla u_{\eta})_{\Omega} \\ &\leq ||u - u_{\eta}||_{\Omega} \cdot ||f + \operatorname{div} q||_{\Omega} + |u - u_{\eta}|_{1,\Omega} \cdot ||q - \nabla u_{\eta}||_{\Omega} \,. \end{aligned}$$

Applying (51) to the above, we have $|u - u_{\eta}|_{1,\Omega}^2 \leq (C_{4,\eta}\eta \|f + \operatorname{div} q\|_{\Omega} + \|q - \nabla u_{\eta}\|_{\Omega})|u - u_{\eta}|_{1,\Omega}$, and hence

$$|u - u_{\eta}|_{1,\Omega} \le C_{4,\eta} \eta \, \|f + \operatorname{div} q\|_{\Omega} + \|q - \nabla u_{\eta}\|_{\Omega} \,.$$
(53)

Here, the constant $C_{4,\eta}$ appears again, and this estimate becomes an a posteriori one, provided that q is specified somehow. The most elegant but quite a restrictive choice is based on the hypercircle method [12, 20], where q is chosen so that f + div q = 0 and hence the use of $C_{4,\eta}$ becomes unnecessary. More common and practical approach is to obtain q by post-processing of u_{η} , for example, by averaging or smoothing ∇u_{η} so as to belong to $H(\text{div}; \Omega)$. For this approach to be really effective, it is at least necessary that $||q - \nabla u_{\eta}||_{\Omega} = O(\eta)$, and preferably $||f + \text{div } q||_{\Omega} = o(\eta)$. Combining (53) with (45) as in (51), we can also obtain a kind of a posteriori L_2 -error estimate as

$$\|u - u_{\eta}\|_{\Omega} \le C_{4,\eta}^2 \eta^2 \|f + \operatorname{div} q\|_{\Omega} + C_{4,\eta} \eta \|q - \nabla u_{\eta}\|_{\Omega} .$$
(54)

In summary, we can effectively utilize error constants such as $C_{4,\eta}$ also in a posteriori error estimation of finite element solutions.

4 Dependence of $C_i(\alpha)$ on α

Up to now, we have given some basic results for dependence of error constants on h, α and θ . In this section, we will consider the dependence of such constants on $\alpha > 0$ in the special case when $\theta = \pi/2$ and h = 1. Actually, we need their behaviors in the range $0 < \alpha \le 1$, and, in view of (32), we want to find their maxima or nice upper bounds there. Furthermore, the limiting case $\alpha \rightarrow +0$ is of some practical interests in the so-called anisotropic mesh refinements [1, 13].

4.1 Definitions and notations

Since each $C_i(\alpha) = C_i(\alpha, \pi/2, 1)$ is defined through minimization of a Rayleigh quotient in terms of norms and/or seminorms over T_α (see (18) through (20)), it is natural to introduce the following transformation $\xi = \Psi_\alpha(x)$ between $x = \{x_1, x_2\} \in T_\alpha$ and $\xi = \{\xi_1, \xi_2\} \in T$:

$$\xi_1 = x_1, \ \xi_2 = x_2/\alpha,$$
 (55)

together with the associated transformation $\tilde{v} = v \circ \Psi_{\alpha}^{-1}$ between functions v over T_{α} and \tilde{v} over T: $\tilde{v}(\xi) = v(x) = v(\xi_1, \alpha\xi_2)$. Notice that $\Psi_{\alpha} = \Psi_{\alpha,\pi/2}$ for $\Psi_{\alpha,\theta}$ in (25).

Then we have the following expressions to (semi-)norms for T_{α} in terms of those for T:

$$\|v\|_{T_{\alpha}}^{2} = \alpha \|\tilde{v}\|_{T}^{2},$$
(56)

$$|v|_{1,T_{\alpha}}^{2} = \alpha a_{\alpha}^{(1)}(\tilde{v}); \quad a_{\alpha}^{(1)}(\tilde{v}) := \left\| \frac{\partial \tilde{v}}{\partial \xi_{1}} \right\|_{T}^{2} + \alpha^{-2} \left\| \frac{\partial \tilde{v}}{\partial \xi_{2}} \right\|_{T}^{2}, \tag{57}$$

$$|v|_{2,T_{\alpha}}^{2} = \alpha a_{\alpha}^{(2)}(\tilde{v}); \ a_{\alpha}^{(2)}(\tilde{v}) := \left\| \frac{\partial^{2} \tilde{v}}{\partial \xi_{1}^{2}} \right\|_{T}^{2} + 2\alpha^{-2} \left\| \frac{\partial^{2} \tilde{v}}{\partial \xi_{1} \partial \xi_{2}} \right\|_{T}^{2} + \alpha^{-4} \left\| \frac{\partial^{2} \tilde{v}}{\partial \xi_{2}^{2}} \right\|_{T}^{2},$$
(58)

where, for example in (56), $v \in L_2(T_\alpha)$ and $\tilde{v} \in L_2(T)$ with $v = \tilde{v} \circ \Psi_\alpha$. By using these α -dependent quadratic forms, the Rayleigh quotients $R_\alpha^{(i)}(v) = R_{\alpha,\pi/2}^{(i)}(v)$ $(0 \le i \le 5)$ for $R_{\alpha,\theta}^{(i)}$'s in (18) through (20) are expressed as

$$R_{\alpha}^{(i)}(v) = \tilde{R}_{\alpha}^{(i)}(\tilde{v}) := \frac{a_{\alpha}^{(1)}(\tilde{v})}{\|\tilde{v}\|_{T}^{2}}; \ v \in V_{\alpha}^{i} \setminus \{0\}, \ \tilde{v} = v \circ \Psi_{\alpha}^{-1} \in V^{i} \setminus \{0\} \ (0 \le i \le 3), \ (59)$$

$$R_{\alpha}^{(4)}(v) = \tilde{R}_{\alpha}^{(4)}(\tilde{v}) := \frac{a_{\alpha}^{(2)}(\tilde{v})}{a_{\alpha}^{(1)}(\tilde{v})}; \quad v \in V_{\alpha}^{4} \setminus \{0\}, \quad \tilde{v} = v \circ \Psi_{\alpha}^{-1} \in V^{4} \setminus \{0\},$$
(60)

$$R_{\alpha}^{(5)}(v) = \tilde{R}_{\alpha}^{(5)}(\tilde{v}) := \frac{a_{\alpha}^{(2)}(\tilde{v})}{\|\tilde{v}\|_{T}^{2}}; \ v \in V_{\alpha}^{4} \setminus \{0\}, \ \tilde{v} = v \circ \Psi_{\alpha}^{-1} \in V^{4} \setminus \{0\}.$$
(61)

We can now analyze the constants $C_i(\alpha)$'s over the common triangle T, at the expense of explicit appearance of the parameter α in the Rayleigh quotients.

We also present the bilinear forms associated to the quadratic forms $a_{\alpha}^{(i)}(\cdot)$'s for i = 1, 2:

$$a_{\alpha}^{(1)}(u,v) := \left(\frac{\partial u}{\partial x_1}, \frac{\partial v}{\partial x_1}\right)_T + \alpha^{-2} \left(\frac{\partial u}{\partial x_2}, \frac{\partial v}{\partial x_2}\right)_T \qquad ; \forall u, v \in H^1(T),$$

$$(62)$$

$$a_{\alpha}^{(2)}(u,v) := \left(\frac{\partial^2 u}{\partial x_1^2}, \frac{\partial^2 v}{\partial x_1^2}\right)_T + 2\alpha^{-2} \left(\frac{\partial^2 u}{\partial x_1 \partial x_2}, \frac{\partial^2 v}{\partial x_1 \partial x_2}\right)_T + \alpha^{-4} \left(\frac{\partial^2 u}{\partial x_2^2}, \frac{\partial^2 v}{\partial x_2^2}\right)_T \\ ; \forall u, v \in H^2(T) .$$
(63)

Here, for simplicity, we use u and v instead of \tilde{u} and \tilde{v} , and the variable is denoted by $x = \{x_1, x_2\}$ instead of $\xi = \{\xi_1, \xi_2\}$.

The following function spaces will play important roles later:

$$H^{k,Z}(T) = \{ v \in H^k(T) ; \ \partial v / \partial x_2 = 0 \} \quad (k = 1, 2) ,$$
(64)

$$V^{i,Z} = \{ v \in V^i \, ; \, \partial v / \partial x_2 = 0 \} \qquad (0 \le i \le 4) \,, \tag{65}$$

which are actually identified with the spaces of functions dependent only on the variable x_1 as we will see later. By considering bilinear forms $a^{(i)}(\cdot, \cdot)$ for i = 1, 2 over the above type of function spaces, we are naturally led to the following bilinear forms:

$$a_Z^{(1)}(u,v) := \left(\frac{\partial u}{\partial x_1}, \frac{\partial v}{\partial x_1}\right)_T \quad ; \forall u, v \in H^1(T),$$
(66)

$$a_Z^{(2)}(u,v) := \left(\frac{\partial^2 u}{\partial x_1^2}, \frac{\partial^2 v}{\partial x_1^2}\right)_T \quad ; \forall u, v \in H^2(T).$$
(67)

Although these are defined over the whole H^1 and H^2 spaces for convenience, the partial derivatives above can be actually replaced with the ordinary ones when they are considered over the respective $H^{1,Z}$ and $H^{2,Z}$ spaces.

As a characterization of the above $H^{1,Z}(T)$, let us state a fundamental lemma to be used for our analysis. Its proof is omitted here since it can be performed by slightly modifying that for Theorem 3.1.4' of [16]. Of course, essentially the same conclusions are drawn for other spaces in (64) and (65).

Lemma 1. Any $v \in H^{1,Z}(T)$ can be identified with a function v^* of single variable x_1 :

$$v(x_1, x_2) = v^*(x_1)$$
 for a.e. $x = \{x_1, x_2\} \in T$. (68)

Remark 6. The present lemma does not necessarily hold for general domains. It holds for a domain $\Omega \subset \mathbf{R}^2$ which is "connected in x_2 direction" in the sense : For any two points x and x^* in Ω with a common x_1 component, the segment connecting these points is contained in Ω .

4.2 Monotonicity and upper bounds of $C_i(\alpha)$

We first derive some fundamental results for $C_i(\alpha)$'s for $0 < \alpha \le 1$, especially for their upper bounds. With this regard, we owe much the following results to the analysis by Babuška and Aziz [5]. In particular, the estimation $C_4(\alpha) \le C_1$ below is an important consequence derived in [5] and also in [25, 30], and so we here call C_1 the Babuška-Aziz constant. **Theorem 4.** $C_i(\alpha) = C_i(\alpha, \pi/2, 1)$ $(0 \le i \le 5)$ are continuous positive-valued functions of $\alpha \in [0, +\infty[$ (here we consider also for $\alpha > 1$). In addition, except for i = 4, they are monotonically increasing in α . Thus, in particular,

$$C_i(\alpha) \le C_i (= C_i(1)) ; \forall \alpha \in]0,1] \ (i = 0, 1, 2, 3, 5).$$
 (69)

On the other hand, it holds for i = 4 that

$$C_4(\alpha) \le \max\{C_1(\alpha), C_2(\alpha)\} \le C_1(=C_2); \ \forall \alpha \in]0, 1].$$
(70)

Remark 7. It is also possible to show the continuity of constants $C_i(\alpha, \theta)$'s with respect to $\{\alpha, \theta\} \in]0, +\infty[\times]0, \pi[$ by slightly generalizing the arguments below. For $C_4(\alpha)$, bounding (70) assures that it is bounded from above by a monotonically increasing function of α . Moreover, numerical results suggest that it is also monotonically increasing as will be seen later, although we do not have fully theoretical results at present. Existence of $C_i(+0)$ for $i \neq 4$ is clear from the monotonicity stated above, although we will make more detailed analysis in the subsequent subsection including the case i = 4.

Proof. We just give sketches since the arguments employed here are rather standard. As was mentioned in Section 4.1 and also used in [5], we consider the Rayleigh quotients $\tilde{R}_{\alpha}^{(i)}$'s for functions over the common domain T.

For the continuity, we first note that each Rayleigh quotient for a fixed $\tilde{v} \neq 0$ is a continuous positive function of α , so that its infimum over all \tilde{v} is uniformly bounded over any compact interval for α of the form $[\alpha_1, \alpha_2]$; $0 < \alpha_1 < \alpha_2 < +\infty$. It is also clear that the infimum for each $\alpha > 0$ is actually the minimum and cannot be zero (i. e., it is positive), as is shown by the usual arguments based on the Rellich compactness theorem and the reduction to absurdity. Then we can assure the existence of both $\overline{\lim}_{\beta\to\alpha} C_i^{-2}(\beta)$ ($\leq C_i^{-2}(\alpha)$) and $\underline{\lim}_{\beta\to\alpha} C_i^{-2}(\beta)$ for each $\alpha > 0$ and i; $0 \leq i \leq 5$. Choosing an appropriate bounded sequence in V^i associated to the above lower limit, we can prove $C_i^{-2}(\alpha) \leq \underline{\lim}_{\beta\to\alpha} C_i^{-2}(\beta)$, i. e., the continuity at α , by adopting the weakly lower semi-continuity of the numerator and the continuity of the denominator appearing in the definition of $\tilde{R}_{\alpha}^{(i)}$ with respect to the metric of V^i . Here, the Rellich type compactness theorem is again needed, and arguments similar to those in the subsequent subsection are used as well.

For the monotonicity and (70), we omit the proof since they can be concluded in completely the same fashion as in [5]. \Box

4.3 Asymptotic behaviors of constants as $\alpha \rightarrow +0$

We will now analyze the asymptotic behaviors of the constants $C_i(\alpha)$'s $(0 \le i \le 5)$ as $\alpha \to +0$ by adopting various techniques developed e.g. in [23]. In particular, the right limit values $C_i(+0)$'s are given by zeros of certain transcendental equations (derived from eigenvalue problems of ordinary differential equations, ODE's) in terms of the hypergeometric functions [32]. For example, $C_2(+0)^{-1}$ is equal to the first positive zero of the Bessel function $J_0(\cdot)$. Moreover, these right limits give lower bounds for respective $C_i(\alpha)$'s, including the non-trivial case i = 4. Such results can be of use for understanding and analyzing the so-called "anisotropic triangulations" discussed e.g. in [1, 8, 13].

4.3.1 Main results

We first present the main results as a theorem below.

Theorem 5. For each i $(0 \le i \le 5)$, $C_i(+0) = \lim_{\alpha \to +0} C_i(\alpha)$ exists and is positive. Moreover, they are the lower limits of the respective constants, i.e., $C_i(+0) = \inf_{\alpha>0} C_i(\alpha)$ for $0 \le i \le 5$. They are characterized by the relations $C_i(+0) = 1/\sqrt{\lambda^{(i)}}$ for $0 \le i \le 5$, where $\lambda^{(i)}$'s are the minimum eigenvalues of the following eigenvalue problems:

 $0 \leq i \leq 3$: Find $\lambda(=\lambda^{(i)}) \in \mathbf{R}$ and $u \in V^{i,Z} \setminus \{0\}$ such that

$$a_Z^{(1)}(u,v) = \lambda(u,v)_T \; ; \; \forall v \in V^{i,Z} \; , \tag{71}$$

i = 4: Find $\lambda(=\lambda^{(4)}) \in \mathbf{R}$ and $u \in V^{4,Z} \setminus \{0\}$ such that

$$a_Z^{(2)}(u,v) = \lambda a_Z^{(1)}(u,v) \; ; \; \forall v \in V^{4,Z} \; , \tag{72}$$

i = 5: Find $\lambda(=\lambda^{(5)}) \in \mathbf{R}$ and $u \in V^{4,Z} \setminus \{0\}$ such that

$$a_Z^{(2)}(u,v) = \lambda(u,v)_T \; ; \; \forall v \in V^{4,Z} \; .$$
(73)

These eigenvalue problems are also expressed by those for the following 2nd- or 4th-order ordinary differential equations for u = u(s) over the interval [0, 1].

$$i = 0: -[(1-s)u'(s)]' = \lambda^{(0)}(1-s)u(s) \quad (0 < s < 1), \ \int_0^1 (1-s)u(s) \, ds = u'(0) = 0, \ (74)$$

$$i = 1: -[(1-s)u'(s)]' = \lambda^{(1)}(1-s)u(s) + C \quad (0 < s < 1), \ \int_0^1 u(s) \, ds = u'(0) = 0, \ (75)$$

$$i = 2: \qquad -[(1-s)u'(s)]' = \lambda^{(2)}(1-s)u(s) \quad (0 < s < 1), \ u(0) = 0, \qquad (76)$$

i = 3: essentially the same as for i = 1;

$$-[(1-s)u'(s)]' = \lambda^{(3)}(1-s)u(s) + C \quad (0 < s < 1), \quad \int_0^1 u(s) \, ds = u'(0) = 0, \quad (77)$$

i = 4: actually reduces to the case i = 1;

$$[(1-s)u''(s)]'' = -\lambda^{(4)}[(1-s)u'(s)]' \quad (0 < s < 1), \ u(0) = u(1) = u''(0) = 0,$$
(78)

$$i = 5: [(1-s)u''(s)]'' = \lambda^{(5)}(1-s)u(s) \quad (0 < s < 1), \ u(0) = u(1) = u''(0) = 0.$$
(79)

Here, C is an unknown constant to be determined simultaneously with u and $\lambda^{(i)}$ (i = 1, 3).

Remark 8. In (74), the two conditions $\int_0^1 (1 - s)u(s) ds = 0$ and u'(0) = 0 are actually identical for smooth u as may be seen by integrating the differential equation in (74) from s = 0 to s = 1. In the above, the numbers of boundary conditions are smaller than the orders of differential equations. This is mainly because the ordinary differential equations above have singularities in their coefficients at s = 1, so that the usual full numbers of boundary conditions are excessive to decide eigenfunctions in respective spaces V^j 's $(0 \le j \le 4)$. The eigenfunctions and eigenvalues are determined by using the hypergeometric functions [32], and the results are summarized in Appendix.

4.3.2 **Proof of main results**

Let us now prove Theorem 5. The statements for us to show range from i = 0 to i = 5, but the methods and techniques to be employed are more or less alike. Among them, the analysis for i = 4 appears to be the most complicated, so that we will present the proof almost exclusively in this particular case. We will proceed in several steps.

1] To analyze asymptotic behaviors of $C_4(\alpha)$, let us define $\lambda_4(\alpha)$ for $\alpha > 0$ by $\lambda_4(\alpha) := C_4^{-2}(\alpha) > 0$, that is,

$$\lambda_4(\alpha) = \inf_{v \in V^4 \setminus \{0\}} \tilde{R}^{(4)}_{\alpha}(v); \quad \tilde{R}^{(4)}_{\alpha}(v) = \frac{a^{(2)}_{\alpha}(v)}{a^{(1)}_{\alpha}(v)}, \tag{80}$$

where, for simplicity, v is used instead of \tilde{v} unlike in (60). By the standard arguments, the infimum is shown to be actually the minimum, and is attained by a certain $u \in V^4 \setminus \{0\}$.

Moreover, $\{\lambda_4(\alpha), u\}$ is an eigenpair of the following eigenvalue problem:

$$a_{\alpha}^{(2)}(u,v) = \lambda_4(\alpha) a_{\alpha}^{(1)}(u,v); \quad \forall v \in V^4 ,$$
 (81)

where $a_{\alpha}^{(i)}(u, v)$ for i = 1, 2 are the bilinear forms associated to $a_{\alpha}^{(i)}(\cdot)$'s, see (62) and (63). Of course, the present $\lambda_4(\alpha) > 0$ is the minimum eigenvalue of (81).

Since $\tilde{R}^{(4)}_{\alpha}(v)$ is a homogeneous form of order 0, we can normalize the eigenfunction u as

$$a_{\alpha}^{(1)}(u) = 1.$$
(82)

2] Let us show that $\lambda_4(+0) = \lim_{\alpha \to +0} \lambda_4(\alpha)$ exists and is positive. Taking $v \in V^4 \setminus \{0\}$ in (80) as $v(x_1, x_2) = x_1(1 - x_1)$, we can see that $\lambda_4(\alpha)$ is uniformly bounded for $\alpha \in]0, \infty[$, and hence $\alpha = 0$ is an accumulation point. In particular, both $\lambda_4^* := \underline{\lim}_{\alpha \to +0} \lambda_4(\alpha) \ge 0$ and $\lambda_4^{\dagger} := \overline{\lim}_{\alpha \to +0} \lambda_4(\alpha)$ exist. Then we can find a sequence $\{\alpha_n\}_{n=1}^{\infty}$ in]0, 1] such that

$$\lim_{n \to \infty} \alpha_n = 0, \quad \lim_{n \to \infty} \lambda_4(\alpha_n) = \lambda_4^*.$$
(83)

We must show that λ_4^* coincides with λ_4^{\dagger} to conclude the existence of the right limit $\lambda_4(+0)$.

Associated to the above sequence $\{\alpha_n\}$, there exists a sequence $\{u_n\}$ in $V^4 \setminus \{0\}$ such that each member satisfies (81) and (82), i. e., $a_{\alpha_n}^{(1)}(u_n) = 1$, $a_{\alpha_n}^{(2)}(u_n) = \lambda_4(\alpha_n)$, and

$$a_{\alpha_n}^{(2)}(u_n, v) = \lambda_4(\alpha_n) a_{\alpha_n}^{(1)}(u_n, v) ; \quad \forall v \in V^4 .$$
(84)

Since $|u|_{1,T}^2 = \sum_{i=1}^2 \|\partial u / \partial x_i\|_T^2 \le a_{\alpha}^{(1)}(u)$ and $|u|_{2,T}^2 = \sum_{i,j=1}^2 \|\partial^2 u / \partial x_i \partial x_j\|_T^2 \le a_{\alpha}^{(2)}(u)$ for $\alpha > 0$, we have for $\{u_n\}$ that

$$|u_n|_{1,T}^2 + |u_n|_{2,T}^2 \le 1 + \lambda_4(\alpha_n) \quad (n = 1, 2, ...).$$
(85)

That is, $\{u_n\}$ is bounded with respect to the semi-norms of $H^1(T)$ and $H^2(T)$ appearing above. Moreover, we can show that $\{||u_n||_T\}$ is also bounded by noting that $\{u_n\}$ is a sequence in V^4 and utilizing the Rellich theorem. Thus $\{u_n\}$ is a bounded sequence in $H^2(T)$, so that there exists a subsequence of $\{u_n\}$, again denoted by $\{u_n\}$ for convenience, and $u_0 \in V^4$ such that, for $n \to \infty$,

$$u_n \to u_0$$
 weakly in $V^4 \subset H^2(T)$, and strongly in $H^1(T)$, (86)

where the strong convergence is concluded by the Rellich theorem. Substituting $v_Z \in V^{4,Z}$ as $v \in V^4$ in (84) and then taking the limit for $n \to \infty$, we find

$$a_Z^{(2)}(u_0, v_Z) = \lambda_4^* a_Z^{(1)}(u_0, v_Z) \, ; \, \forall v_Z \in V^{4, Z} \, .$$
(87)

Furthermore, since $\|\partial u_n/\partial x_2\|_T^2 = \alpha_n^2 \left[a_{\alpha_n}^{(1)}(u_n) - \|\partial u_n/\partial x_1\|_T^2\right]$ from (62), we can show that

$$\partial u_0 / \partial x_2 = 0$$
, i.e., $u_0 \in V^{4,Z}$. (88)

Thus we have obtained (72), provided that $u_0 \neq 0$. For the moment, we cannot exclude the possibility that $u_0 = 0$, so that we will now consider the two cases below.

3] (Case: $u_0 \neq 0$) In this case, $\{\lambda_4^*, u_0\} \in \mathbf{R} \times V^{4,Z}$ is an eigenpair of (87), and is also associated with the following minimization problem:

$$\lambda = \inf_{v \in V^{4,Z} \setminus \{0\}} \frac{a_Z^{(2)}(v)}{a_Z^{(1)}(v)} \,. \tag{89}$$

It is not difficult to show that this minimization problem has a minimum $\mu_4 > 0$, which is at the same time the minimum eigenvalue of (87) and whose arbitrary minimizer $v^Z \in V^{4,Z} \setminus \{0\}$ is an associated eigenfunction. Noting that λ_4^* is an eigenvalue of (87) and

$$\mu_{4} = \inf_{v \in V^{4,Z} \setminus \{0\}} \frac{a_{Z}^{(2)}(v)}{a_{Z}^{(1)}(v)} = \frac{a_{Z}^{(2)}(v^{Z})}{a_{Z}^{(1)}(v^{Z})} = \frac{a_{\alpha}^{(2)}(v^{Z})}{a_{\alpha}^{(1)}(v^{Z})} = \tilde{R}_{\alpha}^{\{4\}}(v^{Z}) \ge \lambda_{4}(\alpha) \quad \text{for } \forall \alpha \in]0, \infty[,$$
(90)

we have $\mu_4 \leq \lambda_4^* = \underline{\lim}_{\alpha \to +0} \lambda_4(\alpha) \leq \lambda_4^{\dagger} = \overline{\lim}_{\alpha \to +0} \lambda_4(\alpha) \leq \mu_4$, that is, λ_4^* coincides with λ_4^{\dagger} and also with μ_4 , so that it is the minimum eigenvalue of (87) and u_0 is an associated eigenfunction. Thus, if $u_0 \neq 0$ for all possible subsequences, λ_4^* is uniquely determined independently of the sequences like original $\{u_n\}$, so that the present λ_4^* is the true right limit $\lambda_4(+0)$. Furthermore, from the above consideration, λ_4^* is also the upper limit of $\lambda_4(\alpha)$ for $\alpha \in]0, \infty[$, that is, $1/\sqrt{\lambda_4^*}$ is the lower limit of $C_4(\alpha)$. Of course, such conclusions are justified provided that u_0 cannot be 0. By using $v(x_1, x_2) = \sin \pi x_1 \in V^{4,Z} \setminus \{0\}$ in the Rayleigh quotient appearing in (89), we can also show that

$$0 < \mu_4 = \lambda_4^* \le \pi^2 < 10.$$
⁽⁹¹⁾

4] (Case : $u_0 = 0$) Let us define w_n by $w_n = \alpha_n^{-1} \partial u_n / \partial x_2$ (n = 1, 2, ...). Then we can see that $w_n \in V^2 \subset H^1(T)$. Since $u_n \to u_0 = 0$ strongly in $H^1(T)$ and $a_{\alpha_n}^{(1)}(u_n) = 1$, it holds that $\|w_n\|_T^2 = 1 - \|\partial u_n / \partial x_1\|_T^2 \to 1$. Moreover, $a_{\alpha_n}^{(2)}(u_n) = \lambda_4(\alpha_n)$, i. e.,

$$\|\partial^2 u_n / \partial x_1^2\|_T^2 + 2\|\partial w_n / \partial x_1\|_T^2 + \alpha_n^{-2}\|\partial w_n / \partial x_2\|_T^2 = \lambda_4(\alpha_n) \quad (n = 1, 2, ...),$$
(92)

is uniformly bounded, so that $\{w_n\}$ is bounded in $H^1(T)$ and $\|\partial w_n/\partial x_2\|_T \to 0 \ (n \to \infty)$. Thus, further choosing a subsequence of $\{w_n\}$ and denoting it by the same notation for simplicity, we can show the existence of $w_0 \in V^{2,Z} \setminus \{0\}$ with $\|w_0\|_T = 1$ such that, for $n \to \infty$,

$$w_n \to w_0$$
 weakly in $V^2 \subset H^1(T)$, and strongly in $L^2(T)$. (93)

Let v^* be an arbitrary function of x_1 such that $v^* \in C^2([0,1])$ with $v^*(0) = 0$, and take $v \in V^4$ in (84) as $v(x_1, x_2) = v^*(x_1)x_2$. For simplicity, we will identify v^* with $v^* \otimes 1_{x_2}$, where 1_{x_2} is the unit constant function of x_2 . Then (84) becomes

$$\alpha_n \left(\frac{\partial^2 u_n}{\partial x_1^2}, \frac{\partial^2 v}{\partial x_1^2}\right)_T + 2 \left(\frac{\partial w_n}{\partial x_1}, \frac{\partial v^*}{\partial x_1}\right)_T = \lambda_4(\alpha_n) \left[\alpha_n \left(\frac{\partial u_n}{\partial x_1}, \frac{\partial v}{\partial x_1}\right)_T + (w_n, v^*)_T\right].$$
(94)

Letting $n \to \infty$ above, we find that $w_0 \in V^{2,Z} \setminus \{0\}$ satisfies

$$2\left(\frac{\partial w_0}{\partial x_1}, \frac{\partial v^*}{\partial x_1}\right)_T = \lambda_4^*(w_0, v^*)_T, \quad \text{i. e.,} \quad a_Z^{(1)}(w_0, v^*) = \frac{1}{2}\lambda_4^*(w_0, v^*)_T.$$
(95)

Moreover, the above holds even for $\forall v^*$ taken from $V^{2,Z}$, since any functions in $V^{2,Z}$ can be approximated by C^2 functions of x_1 vanishing at $x_1 = 0$. Thus the present relation can be viewed as an eigenvalue problem which has $\{\lambda_4^*/2, w_0\}$ as an eigenpair. As usual, we can show that all the eigenvalues are positive, so that $\lambda_4^* > 0$.

By Lemma 1, w_0 can be identified with a function w^* of a single variable x_1 , so that (95) can be expressed by

$$\int_{0}^{1} (1-x_1) \frac{dw^*}{dx_1}(x_1) \frac{dv^*}{dx_1}(x_1) \, dx_1 = \frac{1}{2} \lambda_4^* \int_{0}^{1} (1-x_1) w^*(x_1) v^*(x_1) \, dx_1 \,. \tag{96}$$

Taking v^* from $C_0^{\infty}(]0,1[)$, we have, in the distributional sense (and actually in the classical sense as well) on the interval]0,1[,

$$-\frac{d}{dx_1}\left[(1-x_1)\frac{dw^*}{dx_1}(x_1)\right] = \frac{1}{2}\lambda_4^*(1-x_1)w^*(x_1).$$
(97)

Moreover, it follows from the condition $w_0 \in V^{2,Z}$ that $w^*(0) = 0$. Since $\lambda_4^* > 0$, the general solution of the above is of the form, for arbitrary constants c_1 and c_2 ,

$$w^*(x_1) = c_1 J_0\left(\sqrt{\frac{\lambda_4^*}{2}}(1-x_1)\right) + c_2 Y_0\left(\sqrt{\frac{\lambda_4^*}{2}}(1-x_1)\right) , \qquad (98)$$

where $J_0(\cdot)$ and $Y_0(\cdot)$ are the 0-th order Bessel functions of the first and second kinds, respectively. As is well known, $J_0(\cdot)$ is sufficiently smooth, while $Y_0(\cdot)$ is of the form $Y_0(s) = c_3 \log s + r(s)$ for s > 0, where $c_3 \neq 0$ is a constant and r(s) is a sufficiently smooth remainder term [32]. Consequently, c_2 must be zero for w_0 to belong to $V^{2,Z} \subset H^1(T)$. Then by considering the conditions $w^*(0) = 0$ and $c_1 \neq 0$, $J_0(\sqrt{\lambda_4^*/2})$ must be zero, that is, $\sqrt{\lambda_4^*/2}$ is equal to a positive zero of $J_0(\cdot)$. In fact $J_0(\cdot)$ has countably many positive zeros without any accumulation points except $+\infty$ [32]. Denoting the smallest positive zero by $\gamma_0 > 0$, we have

$$\lambda_4^* \ge 2\gamma_0^2 \,. \tag{99}$$

We can show that $\gamma_0 > 2.25 = 9/4$, so that $\lambda_4^* > 10$. Comparing this with (91), i.e., $10 > \mu_4 \ge \sup_{\alpha>0} \lambda_4(\alpha) \ge \lambda_4^*$, we have a contradiction, and can exclude the possibility that $u_0 = 0$.

Remark 9. Although it is well known that $\gamma_0 = 2.4048...$ numerically, we must verify that $\gamma_0 > 2.25$ for strict analysis. This can be done for example by using the well-known power series

expansion
$$J_0(s) = \sum_{m=0}^{\infty} \frac{1}{(m!)^2} (-\frac{s^2}{4})^m$$
 and numerical verification techniques, cf. [18, 33].

5] We have now proved that λ_4^* and $u_0 \neq 0$ are actually the minimum eigenvalue and the associated eigenfunction of (72), respectively, and that $C_4(+0) = 1/\sqrt{\lambda_4^*} = \inf_{\alpha>0} C_4(\alpha)$.

It is not difficult to prove (78). That is, the differential equation can be obtained just as we derived (97) from (96), while u(0) = u(1) = 0 follow from the condition $u_0 \in V^{4,Z}$. Finally, u''(0) = 0 is obtained as a natural boundary condition associated to (72).

Let us also show that (78) reduces to (75). Denoting u' by v and then integrating the differential equation in (78) with respect to the variable s, we have

$$-[(1-s)v'(s)]' = \lambda^{(4)}[(1-s)v(s)] + C, \qquad (100)$$

which coincides with the differential equation in (75) after rewriting v as u. The boundary condition v'(0) = 0 follows from u''(0) = 0, and the condition $\int_0^1 v(s) ds = 0$ is derived from the relation $\int_0^1 u'(s) ds = u(1) - u(0) = 0$. Once v is determined, u can be reconstructed by integration: $u(s) = \int_0^s v(t) dt$. Consequently, the present case i = 4 reduces to the case i = 1.

6] In the cases other than i = 4, the analyses are a bit easier since the denominators of $\tilde{R}_{\alpha}^{(i)}$'s do not depend on α . For example, (71) and (73) can be derived easily. We just show, in the case of i = 1, how to derive (75) from (71).

For i = 1, $u = u(x_1, x_2)$ and $v = v(x_1, x_2)$ in $V^{1,Z}$ can be identified with functions $u^* = u^*(x_1)$ and $v^* = v^*(x_1)$, respectively, so that (71) for i = 1 can be expressed by, as (96),

$$\int_0^1 (1-x_1) \frac{du^*}{dx_1}(x_1) \frac{dv^*}{dx_1}(x_1) \, dx_1 = \lambda \int_0^1 (1-x_1) u^*(x_1) v^*(x_1) \, dx_1 \,. \tag{101}$$

Let us consider dw^*/dx_1 for $\forall w^* \in C_0^{\infty}(]0,1[)$. Then it can be identified with a function in $V^{1,Z}$, so that, by substituting it into (101) as v^* , we have in the sense of distribution that

$$\frac{d^2}{dx_1^2} \left[(1-x_1) \frac{du^*}{dx_1}(x_1) \right] = -\lambda \frac{d}{dx_1} \left[(1-x_1) u^*(x_1) \right], \tag{102}$$

from which we obtain the differential equation in (75). The condition $\int_0^1 u(s) ds = 0$ follows from $u \in V^{1,Z}$, while u'(0) = 0 is a natural boundary condition associated to (71) for i = 1.

Remark 10. Table 1 shows numerical results for $C_i(+0)$ $(0 \le i \le 5)$ by Mathematica[®], cf. Appendix.

Table 1: Numerical values of $C_i(+0)$ $(0 \le i \le 5)$

i	0	1, 3, 4	2	5
$C_i(+0)$	0.26098	0.32454	0.41583	0.10790

5 A posteriori estimation of some constants

It is in general very difficult to determine exact values of various constants defined in Section 2 for $T_{\alpha,\theta}$ of general shape. Numerically, we can adopt the FEM to obtain approximate values to such constants as may be found e.g. in [4, 7, 22, 30], but their quantitative error estimates are often unavailable. In this section, as an application of our results, let us give a kind of a posteriori estimation of $C_i(\alpha, \theta)$'s $(0 \le i \le 3)$ by adopting the P_1 (piecewise linear) FEM. At present, our approach gives only approximate or numerical boundings of constants, but they can be turned into mathematically correct boundings provided that appropriate numerical verification methods are introduced.

Our approach is based on the classical a priori error estimates for the finite element approximations to the smallest non-zero eigenvalue of the (minus) Laplacian with the Neumann or Dirichlet boundary conditions, cf. e. g. Schultz [29].

5.1 Preliminaries

First, let us make some preparations. Let Ω be a bounded convex polygonal domain. In the present applications, it is often the triangular domain $T_{\alpha,\theta}$. Let us also consider a closed linear subspace $H_s^1(\Omega)$ of $H^1(\Omega)$, which can be infinite-dimensional and satisfies

$$H^1_s(\Omega) \neq \{0\}, \quad 1 \notin H^1_s(\Omega),$$
 (103)

where 1 is the constant function of unit value in Ω . A typical example of such $H_s^1(\Omega)$ is $H_0^1(\Omega)$ considered in Section 3.4.

As a generalization of (42), we can consider the problem of finding $u \in H^1_s(\Omega)$, for a given $f \in L_2(\Omega)$, such that

$$(\nabla u, \nabla v)_{\Omega} = (f, v)_{\Omega} \quad (\forall v \in H^1_s(\Omega)).$$
(104)

The uniqueness and existence of u in $H^1_s(\Omega)$ are almost trivial, so that we can define an operator G_s by

$$G_s: f \in L_2(\Omega) \mapsto u \in H^1_s(\Omega) \text{ determined by } (104) .$$
(105)

As a generalization of the problem related to (18), let us also consider a minimization problem for the Rayleigh quotient

$$R^{s}(v) := \frac{|v|_{1,\Omega}^{2}}{\|v\|_{\Omega}^{2}}; \ v \in H^{1}_{s}(\Omega) \setminus \{0\}.$$
(106)

The minimum actually exists and is positive under (103) as may be shown by the compactness arguments. Moreover, denoting the minimum and an associated minimizer by $\lambda > 0$ and $u \in H_s^1(\Omega) \setminus \{0\}$, respectively, they satisfy

$$(\nabla u, \nabla v)_{\Omega} = \lambda(u, v)_{\Omega} \quad (\forall v \in H^1_s(\Omega)).$$
(107)

By using G_s in (104), the present $u \in H^1_s(\Omega)$ is shown to satisfy $u = \lambda G_s u$.

To apply the P_1 FEM to the above two problems, we first introduce a regular family of triangulations $\{\mathcal{T}^{\eta}\}_{\eta>0}$ of Ω as was mentioned in Section 3.4, and then construct the piecewise linear finite element space $S^{\eta} \subset H^1(\Omega)$ for each \mathcal{T}^{η} as

$$S^{\eta} := \{ v_{\eta} \in C(\overline{\Omega}) \mid v_{\eta} | K \text{ is a linear function for each } K \in \mathcal{T}^{\eta} \}.$$
(108)

For $u \in H^2(\Omega) \subset C(\overline{\Omega})$, we can define the piecewise linear interpolant $\Pi^{\eta,1} u \in S^{\eta}$ by

$$(\Pi^{\eta,1}u)(x^*) = u(x^*) \text{ for any vertex } x^* \text{ of } \mathcal{T}^{\eta}.$$
(109)

We will also use the parameters $\eta = \max_{K \in \mathcal{T}^{\eta}} h_K$, $C_{4,\eta} = \max_{K \in \mathcal{T}^{\eta}} C_4(\alpha_K, \theta_K)$ and $C_{5,\eta} = \max_{K \in \mathcal{T}^{\eta}} C_5(\alpha_K, \theta_K)$ defined in Section 3.4. Then we have the following interpolation estimates for the above u as was discussed in Section 3.4 :

$$|u - \Pi^{\eta,1}u|_{1,\Omega} \le C_{4,\eta}\eta |u|_{2,\Omega} , \quad ||u - \Pi^{\eta,1}u||_{\Omega} \le C_{5,\eta}\eta^2 |u|_{2,\Omega} .$$
(110)

To construct approximate problems to (104) and the minimization of (106), let us consider the subspace $S^{\eta,s}$ of S^{η} defined by

$$S^{\eta,s} := S^{\eta} \cap H^1_s(\Omega) , \qquad (111)$$

which we assume to be different from $\{0\}$. Of course, various other finite-dimensional subspaces of $H_s^1(\Omega)$ are available in place of $S^{\eta,s}$, but the above one is theoretically simple and also practically favorable in many cases.

Then an approximation to (104) is to find $u_{\eta} \in S^{\eta,s}$, for a given $f \in L_2(\Omega)$, such that

$$(\nabla u_{\eta}, \nabla v_{\eta})_{\Omega} = (f, v_{\eta})_{\Omega} \quad (\forall v_{\eta} \in S^{\eta, s}).$$
(112)

The uniqueness and existence of u_{η} in $S^{\eta,s}$ are trivial, so that we can define an operator G_s^{η} approximating G_s by

$$G_s^{\eta} : f \in L_2(\Omega) \mapsto u_{\eta} \in S^{\eta,s} \text{ determined by (112)}.$$
(113)

As generalizations of (44) and (45), we have

$$|G_s f - G_s^{\eta} f|_{1,\Omega} = \min_{v_\eta \in S^{\eta,s}} |G_s f - v_\eta|_{1,\Omega} , \qquad (114)$$

$$\|G_s f - G_s^{\eta} f\|_{\Omega} \le |G_s f - G_s^{\eta} f|_{1,\Omega} \inf_{v_{\eta} \in S^{\eta,s}} \sup_{g \in L_2(\Omega) \setminus \{0\}} \frac{|G_s g - v_{\eta}|_{1,\Omega}}{\|g\|_{\Omega}} .$$
(115)

On the other hand, an approximation problem related to $R^s(\cdot)$ is to find its minimum in $S^{\eta,s} \setminus \{0\}$. In this case, the existence of the minimum is again trivial, and the minimum λ^{η} and an associated minimizer $u_{\eta} \in S^{\eta,s} \setminus \{0\}$ satisfies the relation analogous to (107):

$$(\nabla u_{\eta}, \nabla v_{\eta})_{\Omega} = \lambda^{\eta} (u_{\eta}, v_{\eta})_{\Omega} \quad (\forall v_{\eta} \in S^{\eta, s}).$$
(116)

The following results are easy to derive but will play an essential role in our approach, cf. e. g. Theorem 8.3 of [29].

Lemma 2. Let λ and λ^{η} be respectively defined by $\lambda = \min_{v \in H_s^1(\Omega) \setminus \{0\}} R^s(v)$ and $\lambda^{\eta} = \min_{v_\eta \in S^{\eta,s} \setminus \{0\}} R^s(v_\eta)$, and $u \in H_s^1(\Omega)$ be an minimizer associated to λ such that $||u||_{\Omega} = 1$. Then it holds that, for $\forall v_\eta \in S^{\eta,s} \setminus \{0\}$ with $||u - v_\eta||_{\Omega} < 1$,

$$\lambda \le \lambda^{\eta} \le \lambda + \frac{|u - v_{\eta}|_{1,\Omega}^2}{(1 - ||u - v_{\eta}||_{\Omega})^2} \,. \tag{117}$$

The following results are also well known and will be used later, cf. [15].

Lemma 3. Let us consider the problem: for the present Ω and a given $f \in L_2(\Omega)$, find $u \in H^1(\Omega)$ such that

$$(\nabla u, \nabla v)_{\Omega} = (f, v)_{\Omega} \quad (\forall v \in H^{1}(\Omega)).$$
(118)

Then such u exists if and only if

$$(f,1)_{\Omega} = 0$$
, (119)

and is unique up to an additive arbitrary constant function. Moreover, $u \in H^2(\Omega)$ with

$$\|u\|_{2,\Omega} \le \|\Delta u\|_{\Omega} = \|f\|_{\Omega} .$$
(120)

Remark 11. To assure the uniqueness to u, we can for example impose the condition $(u, 1)_{\Omega} = 0$ on u. The present problem corresponds to the one for the Poisson equation with the homogeneous Neumann boundary condition :

$$-\Delta u = f \text{ in } \Omega, \quad \frac{\partial u}{\partial n} = 0 \text{ on } \partial \Omega.$$
 (121)

5.2 A posteriori estimation of $C_0(\alpha, \theta)$

We first give a posteriori estimates to $C_0(\alpha, \theta)$. In this case, $\Omega = T_{\alpha,\theta}$ and $H^1_s(\Omega) = V^0_{\alpha,\theta}$. Let us define an orthogonal projection operator $P^0_{\alpha,\theta} : L_2(T_{\alpha,\theta}) \to L^0_2(T_{\alpha,\theta}) := \{g \in L_2(\alpha,\theta) \mid (g,1)_{T_{\alpha,\theta}} = 0\}$ by

$$P^{0}_{\alpha,\theta}g := g - \frac{\int_{T_{\alpha,\theta}} g(x) \, dx}{\int_{T_{\alpha,\theta}} dx} = g - \frac{(g,1)_{T_{\alpha,\theta}}}{|T_{\alpha,\theta}|} \quad (\forall g \in L_2(T_{\alpha,\theta})) ,$$
(122)

where $|T_{\alpha,\theta}|$ denotes the measure of $T_{\alpha,\theta}$. We can easily show that $P^0_{\alpha,\theta}|H^1(T_{\alpha,\theta})$ is an orthogonal projection operator from $H^1(T_{\alpha,\theta})$ to $V^0_{\alpha,\theta}$ with respect to the standard inner product of $H^1(T_{\alpha,\theta})$: $(u, v)_{1,T_{\alpha,\theta}} := (u, v)_{T_{\alpha,\theta}} + (\nabla u, \nabla v)_{T_{\alpha,\theta}} \ (\forall u, v \in H^1(T_{\alpha,\theta}))$. We also denote the present G_s , G^{η}_s , S^{η} and $S^{\eta,s}$ respectively by $G^0_{\alpha,\theta}$, $G^{\eta,0}_{\alpha,\theta}$, $S^{\eta}_{\alpha,\theta}$. Since $S^{\eta}_{\alpha,\theta}$ contains the constant functions, we find that

$$S^{\eta,0}_{\alpha,\theta} = P^0_{\alpha,\theta} S^\eta_{\alpha,\theta} . \tag{123}$$

From now on, we will omit the subscript $T_{\alpha,\theta}$ for the norms, semi-norms and inner products related to this domain. Noting that $\nabla P^0_{\alpha,\theta} v = \nabla v$ and $(f, P^0_{\alpha,\theta} v) = (P^0_{\alpha,\theta} f, v)$ for $\forall v \in H^1(T_{\alpha,\theta})$, eq. (104) for the present $u \in V^0_{\alpha,\theta}$ becomes

$$(\nabla u, \nabla v) = (P^0_{\alpha,\theta} f, v) \quad (\forall v \in H^1(T_{\alpha,\theta})),$$
(124)

which reduces to (118) under (119). Likewise, eq. (107) for the present $\{\lambda, u\} \in \mathbf{R} \times V^0_{\alpha, \theta} \setminus \{0\}$ becomes

$$(\nabla u, \nabla v) = \lambda(u, v) \quad (\forall v \in H^1(T_{\alpha, \theta})),$$
(125)

since $P^0_{\alpha,\theta}u = u$. By Lemma 3, the above u belongs to $H^2(T_{\alpha,\theta}) \cap V^0_{\alpha,\theta}$ with

$$|u|_2 \le \lambda \|u\| \,. \tag{126}$$

Under the preceding preparations, let us apply Lemma 2 to estimate the minimum eigenvalue $\lambda_{\alpha,\theta}^{\eta,0}$ of (116) in terms of the one $\lambda_{\alpha,\theta}^{0}$ of (107) or (125). The minimizer associated to $\lambda_{\alpha,\theta}^{0}$ is denoted by $u_{\alpha,\theta}^{0}$ with the normalization condition $||u_{\alpha,\theta}^{0}|| = 1$. As v_{η} in (117), we can take various candidates from $S_{\alpha,\theta}^{\eta,0}$. One possibility is to utilize the interpolant $\Pi^{\eta,1}u_{\alpha,\theta}^{0} \in S_{\alpha,\theta}^{\eta}$ of $u_{\alpha,\theta}^{0}$. Unfortunately, it may be outside $S_{\alpha,\theta}^{\eta,0}$, but its projection $P_{\alpha,\theta}^{0}\Pi^{\eta,1}u_{\alpha,\theta}^{0}$ can be used thanks to (123). By taking advantage of properties of the orthogonal projection (122), we find that

$$|u_{\alpha,\theta}^{0} - P_{\alpha,\theta}^{0}\Pi^{\eta,1}u_{\alpha,\theta}^{0}|_{1} = |u_{\alpha,\theta}^{0} - \Pi^{\eta,1}u_{\alpha,\theta}^{0}|_{1}, \qquad (127)$$

$$\|u_{\alpha,\theta}^{0} - P_{\alpha,\theta}^{0}\Pi^{\eta,1}u_{\alpha,\theta}^{0}\| = \|P_{\alpha,\theta}^{0}(u_{\alpha,\theta}^{0} - \Pi^{\eta,1}u_{\alpha,\theta}^{0})\| \le \|u_{\alpha,\theta}^{0} - \Pi^{\eta,1}u_{\alpha,\theta}^{0}\|.$$
(128)

Using (110) and (126), we can evaluate the above in terms of η , $\lambda_{\alpha,\theta}^0$, $C_{4,\eta}$ and $C_{5,\eta}$. Unfortunately, we have not necessarily obtained accurate theoretical upper bounds for $C_{5,\eta}$ as was noted in Section 3.3. So we should try to avoid the use of such a constant.

Another possibility is to use $\tilde{u}^0_{\eta \cdot \alpha, \theta} := \lambda^0_{\alpha, \theta} G^{\eta, 0}_{\alpha, \theta} u^0_{\alpha, \theta}$, which is surely in $S^{\eta, 0}_{\alpha, \theta}$ and is suggested by the identity $u^0_{\alpha, \theta} = \lambda^0_{\alpha, \theta} G^0_{\alpha, \theta} u^0_{\alpha, \theta}$. For this choice, we have

$$|u_{\alpha,\theta}^{0} - \tilde{u}_{\eta,\alpha,\theta}^{0}|_{1} \le |u_{\alpha,\theta}^{0} - P_{\alpha,\theta}^{0}\Pi^{\eta,1}u_{\alpha,\theta}^{0}|_{1} = |u_{\alpha,\theta}^{0} - \Pi^{\eta,1}u_{\alpha,\theta}^{0}|_{1},$$
(129)

$$\|u_{\alpha,\theta}^{0} - \tilde{u}_{\eta,\alpha,\theta}^{0}\| \le |u_{\alpha,\theta}^{0} - \tilde{u}_{\eta,\alpha,\theta}^{0}|_{1} \inf_{v_{\eta} \in S_{\alpha,\theta}^{\eta,0}} \sup_{g \in L_{2}(T_{\alpha,\theta}) \setminus \{0\}} \frac{|G_{\alpha,\theta}^{0}g - v_{\eta}|_{1}}{\|g\|} .$$
(130)

In this case, we only need former part of (110), that is, the values of η , $\lambda_{\alpha,\theta}^0$ and $C_{4,\eta}$, and can actually avoid the use of $C_{5,\eta}$.

Based on the above considerations, we have now the following two a priori error estimates.

Lemma 4 (A priori estimates for $\lambda_{\alpha,\theta}^{\eta,0}$). Let $\lambda_{\alpha,\theta}^{0}$ and $\lambda_{\alpha,\theta}^{\eta,0}$ be defined as above. Then, if $C_{5,\eta}\eta^2\lambda_{\alpha,\theta}^0 < 1$,

$$\lambda_{\alpha,\theta}^{0} \leq \lambda_{\alpha,\theta}^{\eta,0} \leq \lambda_{\alpha,\theta}^{0} + \frac{(C_{4,\eta}\eta\lambda_{\alpha,\theta}^{0})^{2}}{(1 - C_{5,\eta}\eta^{2}\lambda_{\alpha,\theta}^{0})^{2}}.$$
(131)

Similarly, if $C^2_{4,\eta}\eta^2\lambda^0_{\alpha,\theta} < 1$, then

$$\lambda_{\alpha,\theta}^0 \le \lambda_{\alpha,\theta}^{\eta,0} \le \lambda_{\alpha,\theta}^0 + \frac{(C_{4,\eta}\eta\lambda_{\alpha,\theta}^0)^2}{(1 - C_{4,\eta}^2\eta^2\lambda_{\alpha,\theta}^0)^2} \,. \tag{132}$$

Remark 12. In actual application of the above estimates, where the exact value of $C_{4,\eta}$ ($C_{5,\eta}$, resp.) may not be available, we can use an appropriate upper bound $\tilde{C}_{4,\eta}$ ($\tilde{C}_{5,\eta}$, resp.). From the considerations in Section 3.3 for concrete values of these constants, (131) would give a better bounding than (132), if an accurate upper bound $\tilde{C}_{5,\eta}$ of $C_{5,\eta}$ becomes available.

Let us define two functions related to (131) and (132):

$$\varphi_{0,1}(t) := t + \frac{(C_{4,\eta}\eta t)^2}{(1 - C_{5,\eta}\eta^2 t)^2} \qquad \left(0 < t < \frac{1}{C_{5,\eta}\eta^2}\right),\tag{133}$$

$$\varphi_{0,2}(t) := t + \frac{(C_{4,\eta}\eta t)^2}{(1 - C_{4,\eta}^2 \eta^2 t)^2} \qquad (0 < t < \frac{1}{(C_{4,\eta}\eta)^2}),$$
(134)

where t is the variable, while other quantities are considered just parameters. Since these two functions are continuous and monotonically increasing on their domains of definition, they have their inverse functions, which are defined in $]0, +\infty[$ and will be denoted by the popular notations $\varphi_{0,1}^{-1}$ and $\varphi_{0,2}^{-1}$. Of course, these inverse functions are also continuous and monotonically increasing. Then we can easily obtain the following a posteriori estimates or boundings of $\lambda_{\alpha,\theta}^0$ by numerically obtained $\lambda_{\alpha,\theta}^{\eta,0}$.

Theorem 6 (A posteriori estimates for $\lambda_{\alpha,\theta}^0$). Let $\lambda_{\alpha,\theta}^0$, $\lambda_{\alpha,\theta}^{\eta,0}$, $\varphi_{0,1}^{-1}$ and $\varphi_{0,2}^{-1}$ be defined as above. Then it holds that

$$\varphi_{0,1}^{-1}(\lambda_{\alpha,\theta}^{\eta,0}) \le \lambda_{\alpha,\theta}^{0} \le \lambda_{\alpha,\theta}^{\eta,0} \quad \text{if} \ \lambda_{\alpha,\theta}^{\eta,0} < \frac{1}{C_{5,\eta}\eta^2} ,$$
(135)

$$\varphi_{0,2}^{-1}(\lambda_{\alpha,\theta}^{\eta,0}) \le \lambda_{\alpha,\theta}^{0} \le \lambda_{\alpha,\theta}^{\eta,0} \quad \text{if} \ \lambda_{\alpha,\theta}^{\eta,0} < \frac{1}{(C_{4,\eta}\eta)^2} \,. \tag{136}$$

Proof. From the preceding theorem, we have, for example, $(0 <)\lambda_{\alpha,\theta}^0 \le \varphi_{0,1}(\lambda_{\alpha,\theta}^0) \le \varphi_{0,1}(\lambda_{\alpha,\theta}^{\eta,0})$ if $\lambda_{\alpha,\theta}^{\eta,0} < 1/(C_{5,\eta}\eta^2)$. Then (135) follows immediately by operating $\varphi_{0,1}^{-1}$ to this inequality, while (136) can be obtained similarly.

It is now straightforward to obtain boundings to the constant $C_0(\alpha, \theta)$. For example, we have from (135) that

$$1/\sqrt{\lambda_{\alpha,\theta}^{\eta,0}} \le C_0(\alpha,\theta) \le 1/\sqrt{\varphi_{0,1}^{-1}(\lambda_{\alpha,\theta}^{\eta,0})} \quad \text{if } \ \lambda_{\alpha,\theta}^{\eta,0} < \frac{1}{C_{5,\eta}\eta^2} \,. \tag{137}$$

The results (135) and (136) can be also viewed as a posteriori error estimates for $\lambda_{\alpha,\theta}^{\eta,0}$, since (135), for example, can be rewritten as $0 \le \lambda_{\alpha,\theta}^{\eta,0} - \lambda_{\alpha,\theta}^0 \le \lambda_{\alpha,\theta}^{\eta,0} - \varphi_{0,1}^{-1}(\lambda_{\alpha,\theta}^{\eta,0})$.

Remark 13. The results in Lemma 4 and Theorem 6, i. e., estimates (131), (132), (135) and (136), also hold for λ and λ^{η} of Lemma 2 in the case where $H_s^1(\Omega) = H_0^1(\Omega)$ and $S^{\eta,s} = S^{\eta} \cap H_0^1(\Omega)$. In such a case, Lemma 3 cannot be used, but the corresponding G_s has the property $G_s : L_2(\Omega) \to H_0^1(\Omega) \cap H^2(\Omega)$ with $|G_s f|_2 \leq ||f|| \ (\forall f \in L_2(\Omega))$, since Ω is a bounded convex polygonal domain [15]. Moreover, we cannot utilize projection operators like $P_{\alpha,\theta}^0$ above, but, instead, we can take full advantage of the property $\Pi^{\eta,1}G_s f \in S^{\eta,s} \ (\forall f \in L_2(\Omega))$ for the present $\Pi^{\eta,1}$, G_s and $S^{\eta,s}$. The present case is related to the approximation of the Poincaré constant [2], which is essentially the numerical evaluation of the smallest eigenvalue of $-\Delta$ with the homogeneous Dirichlet boundary condition. Similarly, $C_0(\alpha, \theta)$ is associated with the second eigenvalue of $-\Delta$ over $\Omega = T_{\alpha,\theta}$ with the homogeneous Neumann boundary condition, as was noted in Section 2.

5.3 A posteriori estimation of $C_i(\alpha, \theta)$'s (i = 1, 2, 3)

Secondly, we give a posteriori estimates to $C_i(\alpha, \theta)$'s $(1 \le i \le 3)$. In these cases, let us choose or use the notations $\Omega = T_{\alpha,\theta}$, $H_s^1(\Omega) = V_{\alpha,\theta}^i$, $G_s = G_{\alpha,\theta}^i$, $G_s^\eta = G_{\alpha,\theta}^{\eta,i}$, $S^\eta = S_{\alpha,\theta}^\eta$ and $S^{\eta,s} = S_{\alpha,\theta}^{\eta,i}$ for each $i \in \{1,2,3\}$. Let us define an operator $P_{\alpha,\theta}^i : H^1(T_{\alpha,\theta}) \to V_{\alpha,\theta}^i$ $(i \in \{1,2,3\})$ by

$$P^{i}_{\alpha,\theta}v := v - \frac{1}{|e_i|} \int_{e_i} v \, ds \qquad (\forall v \in V^{i}_{\alpha,\theta}) , \qquad (138)$$

where $|e_i|$ denotes the length of edge e_i . Unlike $P^0_{\alpha,\theta}$, the above operators are not well-defined over $L_2(T_{\alpha,\theta})$, but the following relations similar to (123) still hold :

$$S_{\alpha,\theta}^{\eta,i} = P_{\alpha,\theta}^i S_{\alpha,\theta}^{\eta} \quad (1 \le i \le 3) .$$
(139)

Suggested by [26], let us introduce quadratic functions f_i 's $(1 \le i \le 3)$ of $x = \{x_1, x_2\}$ by

$$f_i(x_1, x_2) := \frac{|e_i|}{4|T_{\alpha,\theta}|} [(x_1 - x_1^i)^2 + (x_2 - x_2^i)^2], \qquad (140)$$

where

$$x^{1} = \{x_{1}^{1}, x_{2}^{1}\} = B(\alpha \cos \theta, \alpha \sin \theta), \ x^{2} = \{x_{1}^{2}, x_{2}^{2}\} = A(1, 0), \ x^{3} = \{x_{1}^{3}, x_{2}^{3}\} = O(0, 0).$$
(141)

These functions are sufficiently smooth and satisfy

$$\frac{\partial f_i}{\partial n} = \delta_{ij} \text{ on } e_j \text{ for } \forall i, \forall j \in \{1, 2, 3\}.$$
(142)

Then, for $\forall v \in H^1(T_{\alpha,\theta})$, we find that

$$\int_{e_i} v \, ds = (\nabla f_i, \nabla v) + (\Delta f_i, v) , \qquad (143)$$

so that (138) can be rewritten by

$$P^{i}_{\alpha,\theta}v := v - \frac{1}{|e_i|} [(\nabla f_i, \nabla v) + (\Delta f_i, v)] \qquad (\forall v \in H^1(T_{\alpha,\theta})) .$$
(144)

Similarly to (124), eq. (104) for the present $u \in V^i_{\alpha,\theta}$ becomes

$$(\nabla u, \nabla v) = (f, P^i_{\alpha, \theta} v) \qquad (\forall v \in H^1(T_{\alpha, \theta})), \qquad (145)$$

which can be rewritten by

$$\left(\nabla(u+\frac{(f,1)}{|e_i|}f_i),\nabla v\right) = \left(f-\frac{(f,1)}{|e_i|}\Delta f_i,v\right) \qquad (\forall v \in H^1(T_{\alpha,\theta})).$$
(146)

By Lemma 3, we find that $u + \frac{(f,1)}{|e_i|} f_i \in H^2(T_{\alpha,\theta})$ with

$$\left| u + \frac{(f,1)}{|e_i|} f_i \right|_2 \le \| f - \frac{(f,1)}{|e_i|} \Delta f_i \| , \qquad (147)$$

and hence, by using the triangle and Schwarz inequalities,

$$|u|_{2} \leq ||f|| + \frac{|(f,1)|}{|e_{i}|} (|f_{i}|_{2} + ||\Delta f_{i}||) \leq ||f|| \left[1 + \frac{\sqrt{|T_{\alpha,\theta}|}}{|e_{i}|} (|f_{i}|_{2} + ||\Delta f_{i}||)\right].$$
(148)

Clearly, it holds that

$$|T_{\alpha,\theta}| = \frac{\alpha}{2}\sin\theta , \ |e_1| = 1 , \ |e_2| = \alpha , \ |e_3| = \sqrt{1 + \alpha^2 - 2\alpha\cos\theta} ,$$

$$|f_i|_2 = \frac{\sqrt{2}}{2} ||\Delta f_i|| , \ \Delta f_i(x_1, x_2) = \frac{|e_i|}{|T_{\alpha,\theta}|} ,$$
(149)

so that we have, for $\forall i \in \{1, 2, 3\}$,

$$|u|_{2} \le (2 + \frac{\sqrt{2}}{2}) ||f|| .$$
(150)

Similarly to (145), eq.(107) for the present $\{\lambda, u\} \in \mathbf{R} \times V^i_{\alpha, \theta}$ $(1 \le i \le 3)$ becomes

$$(\nabla u, \nabla v) = \lambda(u, P^i_{\alpha, \theta} v) \qquad (\forall v \in H^1(T_{\alpha, \theta})) .$$
(151)

Thus, we can utilize the results for (145) by taking f in (145) as λu in (151). The approximation problems corresponding to (112) and (116) are also given by using $S_{\alpha,\theta}^{\eta,i}$'s $(1 \le i \le 3)$. Then, just like Lemma 4 and Theorem 6 for $C_0(\alpha, \theta)$, we have the following results for $C_i(\alpha, \theta)$'s $(1 \le i \le 3)$.

Theorem 7 (A priori and a posteriori estimates for $\lambda_{\alpha,\theta}^{\eta,i}$'s $(1 \le i \le 3)$). For each $i \in \{1, 2, 3\}$, let $\lambda_{\alpha,\theta}^{i}$ and $\lambda_{\alpha,\theta}^{\eta,i}$ be respectively the smallest eigenvalues of (107) and (116) in the present case where $H_{s}^{1}(\Omega) = V_{\alpha,\theta}^{i}$ and $S^{\eta,s} = S_{\alpha,\theta}^{\eta,i}$. Then, if $(MC_{4,\eta}\eta)^{2}\lambda_{\alpha,\theta}^{i} < 1$ with $M := 2 + \sqrt{2}/2$, it holds that

$$\lambda_{\alpha,\theta}^{i} \leq \lambda_{\alpha,\theta}^{\eta,i} \leq \lambda_{\alpha,\theta}^{i} + \frac{(MC_{4,\eta}\eta\lambda_{\alpha,\theta}^{i})^{2}}{(1 - M^{2}C_{4,\eta}^{2}\eta^{2}\lambda_{\alpha,\theta}^{i})^{2}}.$$
(152)

and, if $\lambda_{\alpha,\theta}^{\eta,i} < \frac{1}{(MC_{4,\eta}\eta)^2}$, $\varphi_i^{-1}(\lambda_{\alpha,\theta}^{\eta,i}) \le \lambda_{\alpha,\theta}^i \le \lambda_{\alpha,\theta}^{\eta,i}$, (153)

where

$$\varphi_i(t) := t + \frac{(MC_{4,\eta}\eta t)^2}{(1 - M^2 C_{4,\eta}^2 \eta^2 t)^2} \qquad \left(0 < t < \frac{1}{(MC_{4,\eta}\eta)^2} \; ; \; 1 \le i \le 3\right) \; , \tag{154}$$

which is continuous and monotonically increasing.

Remark 14. Because of the factor $M \approx 2.7071...$, efficiency of (152) is worse than that of (132). In the present case, estimates corresponding to (131) and using $C_{5,\eta}$ do not appear to be fully effective unlike in the preceding subsection. This is attributed to the fact that we cannot at present obtain desirable estimates for $||u - P_{\alpha,\theta}^i \Pi^{\eta,1}u||$ ($\forall u \in V_{\alpha,\theta}^i \cap H^2(T_{\alpha,\theta})$; $1 \le i \le 3$), since $P_{\alpha,\theta}^i$ is not definable over $L_2(T_{\alpha,\theta})$ and hence we cannot take advantage of the best approximation property with respect to the L_2 norm.

6 Numerical results

We performed numerical computations to see the actual dependence of various constants on α and θ . Furthermore, we also utilized the obtained exact values or upper bounds of such constants to give quantitative a posteriori error estimates for some eigenvalue problems.

6.1 Computational methods

To obtain approximate values of error constants, we can utilize the FEM quite effectively. In particular, we used the most popular P_1 triangular finite element for numerical computations of $C_i(\alpha, \theta)$'s for $0 \le i \le 3$ by preparing appropriate triangulations of $T_{\alpha,\theta}$. For $C_4(\alpha, \theta)$ and $C_5(\alpha, \theta)$, it is natural to use various triangular finite elements for Kirchhoff plate bending problems, since the associated partial differential equations are of 4th order as is noted in Section 2. In our actual computations, we used the discrete Kirchhoff triangular element presented in [17]. On the other hand, we can also use the Siganevich approach for computation of $C_4(\alpha, \theta)$, which adopts the P_1 triangle and a kind of penalty method for a system of 2nd order partial differential equations similar to the incompressible Stokes system [30]. This method also works well if the penalty parameter is carefully chosen.

In every case, we have a matrix eigenvalue problem as the discretization of the original eigenvalue problem described by a weak form. More specifically, it is a generalized matrix eigenvalue problem with respect to unknown eigenvectors of nodal values of approximate eigenfunctions, and it can be solved for example by the inverse iteration method and the subspace iteration method [9]. A difficulty in deriving such matrix eigenvalue problems is how to deal with linear constraint conditions imposed on the spaces $V_{\alpha,\theta}^i$ for i = 0, 1, 2, 3. Similar constraint conditions are also necessary to deal with, if we compute $C_4(\alpha, \theta)$ by the Siganevich method. On the other hand, we do not have such a difficulty in computing $C_4(\alpha, \theta)$ and $C_5(\alpha, \theta)$ by Kirchhoff type triangular elements, where the linear constraints v(O) = v(A) = v(B) = 0 for $V_{\alpha,\theta}^4$ can be handled as homogeneous "point" conditions.

One possible method is to eliminate some unknown nodal values by using the linear constraints, but then we have non-sparse coefficient matrices in general. Another method is to use the Lagrange multiplier method, which does not essentially destroy the global sparseness of the matrices. We tested both approaches with reasonable results. Various iteration methods may be also available for the same purposes.

The numerical results below are obtained by FEM in the double or quadruple precision arithmetics, without evaluating the errors strictly by the interval analysis. But their accuracy appears to be reasonable at least in graphical level, since finer mesh computations give essentially the same graphs. We hope that effective verification methods will be established in near future, so that numerical results can be of strictly mathematical significance.

6.2 Numerical results for error constants

Here, we first show some results for $C_i(\alpha)$'s $(0 \le i \le 5)$ by the P_1 finite element and the Kirchhoff triangular element in [17] with the uniform triangulation of the domain T_{α} . In such calculations, T_{α} is subdivided into a number of small triangles congruent to $T_{\alpha,\pi/2,h}$ with e.g. h = 1/20. The penalty method in [30] is also tested to calculate $C_4(\alpha)$ approximately.



Figure 2: Numerically obtained graphs for $C_i(\alpha)$'s $(0 \le i \le 5; 0 < \alpha \le 1)$

Figure 2 consists of two parts and illustrates the graphs of approximate $C_i(\alpha)$'s $(0 \le i \le 5)$ versus $\alpha \in]0, 1]$. Exact values of C_0 and $C_1 = C_2$ together with an approximate value of C_5 are also included as horizontal lines in graphs. At $\alpha = 1$, the approximate values coincide well with the available exact ones in Theorem 3, and we can numerically see that $C_1 (= C_2)$ is a nice upper bound of C_4 . For general α , the monotonically increasing behaviors theoretically predicted for $C_i(\alpha)$'s (i = 0, 1, 2, 3, 5) as well as the relation $C_4(\alpha) \le \min\{C_1(\alpha), C_2(\alpha)\}$ are also well observable in the graphs. The present numerical results suggest that $C_4(\alpha)$ is also monotonically increasing, but we have not succeeded in proving such a conjecture. Moreover, when $\alpha \approx 0$, the numerical results agree well with the exact right limits given in Table 1 based on the asymptotic analysis.

For $C_4(\alpha)$, we tested two methods, that is, the P_1 triangle with the penalty method and the Kirchhoff triangle. These two methods turned out to give almost the same results if the meshes are relatively fine and the penalty parameter is appropriately chosen. The graph for $C_4(\alpha)$ in Fig. 2 is actually obtained by the Kirchhoff element, but is indistinguishable in graphical level from the one by the penalty method.

We also performed numerical computations to see the validity and effectiveness of the upper bounds for $C_4(\alpha, \theta)$ given in Corollary 1 and Theorem 2. We here show just one example with $0 < \alpha \leq 1$ and $\theta = 2\pi/3$. That is, we numerically compare $C_4(\alpha, \theta)$, $C_4^{(1)}(\alpha, \theta) := C_4(\alpha)\phi_4(\theta)$ and $C_4^{(2)}(\alpha, \theta) := \nu(\alpha, \theta)/(\sqrt{2}\sin\theta)$ for $\theta = 2\pi/3$, where the latter two functions come from (32) and (35). In the computations, $C_4(\alpha)$, $C_1(\alpha, 2\pi/3)$ and $C_2(\alpha, 2\pi/3)$ were also obtained numerically for use in the above two upper bound formulas, and the uniform meshes were again employed. The results are shown in Fig. 3, and we can see that both $C_4^{(1)}(\alpha, 2\pi/3)$ and $C_4^{(2)}(\alpha, 2\pi/3)$ give upper bounds to $C_4^{(1)}(\alpha, 2\pi/3)$ numerically. Moreover, at least in the present case, $C_4^{(2)}(\alpha, 2\pi/3)$ is superior to $C_4^{(1)}(\alpha, 2\pi/3)$ as an upper bound.

Figures 4 and 5 illustrate numerically obtained contour lines for $C_i(\alpha, \theta)$'s in the $\alpha - \theta$ polar coordinates, where the abscissa denotes $\alpha \cos \theta$, and the ordinate does $\alpha \sin \theta$. The unit circle $\alpha = 1$ is also shown by a dotted curve. The minimum required range for α and θ is specified by (1), but the contour lines are shown for wider ranges, so that we can easily see global behaviors of error constants. These results can be also useful for practical adaptive computations to specify constants in error indicators approximately. Of course, for strict mathematical analysis like numerical verification, we need correct upper bounds to error constants. The contour lines are sometimes cut off in the portions where the expected accuracy may be insufficient. For example, when $\alpha \approx 0$ or $|\theta - \pi/2| \approx \pi/2$, it requires extraordinarily fine meshes to retain sufficient accuracy. The behavior of $C_4(\alpha, \theta)$ appears to be the most complicated among all the constants, and the necessity of the maximum angle condition can be visually recognized. The other constants seem to be uniformly bounded over the unit disk $\alpha \leq 1$.

6.3 A posteriori estimates of eigenvalues

To apply the results in Section 5, let us consider a posteriori estimates or boundings for $C_0 = C_0(1, \pi/2)$ and $C_1 = C_1(1, \pi/2)$ based on the P_1 FEM. We denote the associated eigenvalues by $\lambda_0 = C_0^{-2}$ and $\lambda_1 = C_1^{-2}$, and the results will be shown for these eigenvalues. We can also give a posteriori estimates to general $C_i(\alpha, \theta)$'s for i = 0, 1, 2, 3 by the same approach.



Figure 3: Two upper bounds of $C_4(\alpha, \theta)$ for $\theta = 2\pi/3$

Table 2 gives the boundings for λ_0 based on (135) and (136) of Theorem 6 and those for λ_1 based on (153) of Theorem 7. We tested several meshes, which are uniform ones composed of small triangles similar to the entire domain T as is shown in the same table. The values of parameters $\tilde{C}_{4,\eta}$, $\tilde{C}_{5,\eta}$ and η that are necessary to use the formulas (135), (136) and (153) are specified here as

$$\tilde{C}_{4,\eta} = 0.5, \ \tilde{C}_{5,\eta} = 0.17, \ \eta = 1/N,$$
(155)

where N is the number of elements along each edge of T (N = 4 in the figure of Table 2). Notice here that $\tilde{C}_{4,\eta} = 0.5$ is a theoretical upper bound of $C_{4,\eta}$ (cf. Remark 12), but the above $\tilde{C}_{5,\eta} = 0.17$ is only a numerically obtained approximate upper bound of $C_{5,\eta}$ at present. We tested (135) only to see its effectiveness experimentally.

We can observe that the present simple methods can actually bound C_0 and C_1 from both above and below. As is expected, (135) gives better lower bounds than (136) for coarser meshes. Table 2 also shows that the lower bounds obtained for C_1 are in general rougher than those for C_0 . This is probably attributed to the existence of the factor $M = 2 + \sqrt{2}/2$. Even in this case, we can obtain reasonable results by mesh refinement.

As another application of our method, let us consider the bounding of the first eigenvalue for $-\Delta$ subjected to the Dirichlet boundary condition for the right *n*-polygonal domain Ω_n $(n \ge 3)$, circumscribing the unit disk Ω_{∞} centered at the origin. In this case, the formulas in Lemma 4 and Theorem 6 can be used without modifications as is noted in Remark 13, since each Ω_n is convex. It is well known that the first eigenvalue for Ω_n is monotonically increasing in *n* and is bounded from above by that for Ω_{∞} . The eigenvalues for n = 4 and $n = \infty$ are known as $\pi^2/2$ and the square of the first zero of the Bessel function J_0 , respectively, but it is difficult to determine the exact values for general *n*. So we will numerically evaluate



Figure 4: Contour lines of $C_i(\alpha,\theta)$ for i=0,1,2



Figure 5: Contour lines of $C_i(\alpha, \theta)$ for i = 3, 4, 5

such eigenvalues for several n with a posteriori estimates. At present, such estimates are just numerical, but they will be strictly mathematical estimates when appropriate verification methods become available.

As meshes, we first triangulate the right triangle $\triangle OAB$ with OA = 1, $AB = \tan(\pi/n)$ and $\angle OAB = \pi/2$ just as we did for T and T_{α} in the preceding problems by dividing each edges uniformly into N segments. Then by a reflection and rotations, we can obtain whole meshes for Ω_n , see Fig. 6. Then we can use (136) with

$$\tilde{C}_{4,\eta} = 0.5, \quad \eta = \begin{cases} \sqrt{3}/N & \text{if } n = 3\\ 1/N & \text{if } n \ge 4 \end{cases},$$
(156)

where $\alpha \leq 1$ in all the cases.

The obtained results are summarized in Table 3, from which we can experimentally see the effectiveness of our bounding method.

Table 2: A posteriori estimates for λ_0 and λ_1



 $\eta = 1/N, \quad N = 4$ in the left figure

† Approximate eigenvalue is outside the domain of definition for φ_1^{-1} .

N	bounds for λ_0 by $arphi_{0,1}^{-1}$	bounds for λ_0 by $arphi_{0,2}^{-1}$	bounds for λ_1 by φ_1^{-1}
2	$5.9890 < \lambda_0 < 11.7155$	$6.5550 < \lambda_0 < 11.7155$	$\lambda_1 < 4.3071^{\dagger}$
3	$7.8874 < \lambda_0 < 10.7213$	$8.1463 < \lambda_0 < 10.7213$	$1.9780 < \lambda_1 < 4.2102$
4	$8.7512 < \lambda_0 < 10.3570$	$8.8616 < \lambda_0 < 10.3570$	$2.6006 < \lambda_1 < 4.1713$
8	$9.6055 < \lambda_0 < 9.9946$	$9.6143 < \lambda_0 < 9.9946$	$3.6537 < \lambda_1 < 4.1304$
16	$9.8054 < \lambda_0 < 9.9012$	$9.8060 < \lambda_0 < 9.9012$	$3.9982 < \lambda_1 < 4.1196$
32	$9.8537 < \lambda_0 < 9.8776$	$9.8537 < \lambda_0 < 9.8776$	$4.0864 < \lambda_1 < 4.1168$
64	$9.8656 < \lambda_0 < 9.8716$	$9.8656 < \lambda_0 < 9.8716$	$4.1085 < \lambda_1 < 4.1161$
(∞)	$\lambda_0 = \pi^2 =$	$\lambda_1 \approx 4.115858$	

7 Concluding remarks

We have obtained some explicit relations for the dependence of several interpolation error constants on geometric parameters of triangular finite elements. In particular, we have succeeded in determining some special constants including the Babuška-Aziz constant from very simple



Figure 6: Meshes for *n*-polygonal domains Ω_n with N = 5: n = 3, 4, 5, 10

equations. We can effectively utilize these results to give upper bounds of various a priori and a posteriori error estimates of finite element solutions based on the P_1 and/or P_0 approximate functions. Some numerical results were also given to see the effectiveness of our analysis and the actual behaviors of the error constants. To obtain more clear picture for the dependence of the interpolation error constants, we should also perform various analyses including numerical analysis with verifications, asymptotic analysis etc.

We have mainly considered the conforming P_1 triangle, which can naturally construct subspaces of H^1 space over the entire domain. But there also exists a non-conforming counterpart, which is also based on the piecewise linear polynomials but uses as nodes the midpoints of edges or edges themselves [11, 31]. Analysis of such an element is more complicated, since we must additionally evaluate the errors induced by the interelement discontinuity of the approximate functions. Still we can obtain some results for the interpolation errors as suggested in [19] by using the constants for the P_0 and the conforming P_1 triangles. We will report more refined results to the non-conforming P_1 triangle in subsequent papers.

Acknowledgements The authors would like to express their deepest appreciation to Prof. M.T. Nakao of Graduate School of Mathematical Sciences, Kyushu University and Prof. N. Yamamoto of Department of Computer Science, The University of Electro-Communication for acquainting them with the importance of the present problem and a number of references. Some of the exact computations in this paper were performed with the aid of Mathematica[®] 5. This work was partially supported by Japan Society for the Promotion of Science Grant-in-Aid for Scientific Research (C) (2) 16540096 and (C)(2) 19540115.

n	N	bounds for λ	N	bounds for λ	N	bounds for λ
3	5	$3.9082 < \lambda < 4.4963$	10	$4.2688 < \lambda < 4.4147$	100	$4.3853 < \lambda < 4.3868$
4	5	$4.7700 < \lambda < 5.0211$	10	$4.8954 < \lambda < 4.9569$	100	$4.9344 < \lambda < 4.9351$
5	5	$5.0049 < \lambda < 5.2826$	10	$5.1590 < \lambda < 5.2273$	100	$5.2075 < \lambda < 5.2082$
6	5	$5.1387 < \lambda < 5.4323$	10	$5.3114 < \lambda < 5.3839$	100	$5.3659 < \lambda < 5.3667$
7	5	$5.2220 < \lambda < 5.5257$	10	$5.4078 < \lambda < 5.4831$	100	$5.4666 < \lambda < 5.4674$
8	5	$5.2774 < \lambda < 5.5879$	10	$5.4727 < \lambda < 5.5498$	100	$5.5346 < \lambda < 5.5354$
9	5	$5.3160 < \lambda < 5.6313$	10	$5.5185 < \lambda < 5.5969$	100	$5.5827 < \lambda < 5.5836$
10	5	$5.3440 < \lambda < 5.6628$	10	$5.5520 < \lambda < 5.6313$	100	$5.6181 < \lambda < 5.6190$

Table 3: A posteriori estimates for the first eigenvalue λ associated to Ω_n

References

- [1] G. Acosta and R.G. Durán, "The maximum angle condition for mixed and nonconforming elements," *SIAM Journal on Numerical Analysis*, **37**, 18-36 (2000).
- [2] R.A. Adams and J.J.F. Fournier, Sobolev Spaces, Academic Press, 2003.
- [3] M. Ainsworth and J.T. Oden, *A Posteriori Error Estimation in Finite Element Analysis*, John Wiley & Sons, 2000.
- [4] P. Arbenz, "Computable finite element error bounds for Poisson's equation", IMA J. Numer. Anal., 2, 475-479 (1982).
- [5] I. Babuška and A.K. Aziz, "On the angle condition in the finite element method," *SIAM Journal on Numerical Analysis*, **13**, 214-226 (1976).
- [6] I. Babuška and T. Strouboulis, *The Finite Element Method and Its Reliability*, Clarendon Press, 2001.
- [7] R.E. Bahnhill, J.H. Brown and A.R. Mitchell, "A comparison of finite element error bounds for Poisson's equation", *IMA J. Numer. Anal.*, **1**, 95-103 (1981).
- [8] W. Bangerth and R. Rannacher, *Adaptive Finite Element Methods for Differential Equations*, Birkäuser, 2003.
- [9] K.-J. Bathe, Finite Element Procedures, Prentice-Hall, 1996.
- [10] S.C. Brenner and L.R. Scott, *The Mathematical Theory of Finite Element Methods*, 3rd edn., Springer, 2008.

- [11] P.-G. Ciarlet, The Finite Element Method for Elliptic Problems, SIAM, 2002.
- [12] P. Destuynder and B. Métivet, "Explicit error bounds in a conforming finite element method", *Math. Comp.*, 68, 1379-1396 (1999).
- [13] L. Formaggia and S. Perotto, "New anisotropic a priori error estimates," *Numer. Math.*, 89, 641-667 (2001).
- [14] J.A. Gregory, "Error bounds for linear interpolation on triangles," *The Mathematics of Finite Elements and Applications II (ed. J.R. Whiteman)*, Academic Press, London, p. 163-170, 1976.
- [15] P. Grisvard, Elliptic Problems in Nonsmooth Domains, Pitman, 1985.
- [16] L. Hörmander, *The Analysis of Linear Partial Differential Equations I: Distribution The*ory and Fourier Analysis, Springer, 2003.
- [17] F. Kikuchi, K. Ishii, A locking-free mixed triangular element for the Reissner-Mindlin Plates, in S. N. Atluri, G. Yagawa, T.A. Cruse eds. Computational Mechanics '95 – Theory and Applications. Proc. of the Int. Conf. on Computational Engineering Science, July 30-August 3, 1995, Hawaii, USA, Vol. 2, Springer (1995) pp. 1608-1613.
- [18] F. Kikuchi and X. Liu, "Determination of the Babuška-Aziz constant for the linear triangular finite element", *Japan Journal of Industrial and Applied Mathematics*, 23(1), 75-82 (2006).
- [19] F. Kikuchi and X. Liu, "Estimation of interpolation error constants for the P_0 and P_1 triangular finite elements", *Computer Methods in Applied Mechanics and Engineering*, **196**, 3750-3758 (2007).
- [20] F. Kikuchi and H. Saito, "Remarks on a posteriori error estimation for finite element solutions", J. Comp. Appl. Math., 199, 329-336 (2007).
- [21] P. Knabner and L. Angermann, *Numerical Methods for Elliptic and Parabolic Partial Differential Equations*, Springer, 2003.
- [22] R. Lehmann, "Computable error bounds in finite-element method", *IMA J. Numer. Anal.*, 6, 265-271 (1986).
- [23] J.L. Lions, *Perturbations Singulières dans les Problèmes aux Limites et en Contrôle Optimal*, Springer, 1973.
- [24] M.T. Nakao and N. Yamamoto, *Numerical Methods with Guaranteed Accuracy (in Japanese)*, Nippon-Hyoron-Sha, 1998.
- [25] M.T. Nakao and N. Yamamoto, "A guaranteed bound of the optimal constant in the error estimates for linear triangular element," *Computing [Supplementum*], 15, 163-173 (2001).

- [26] M.T. Nakao and N. Yamamoto, "A guaranteed bound of the optimal constant in the error estimates for linear triangular elements, Part II: Details," *Perspectives on Enclosure Methods (eds. U. Kulisch et al.), the Proceedings Volume for Invited Lectures of SCAN2000*, Springer-Verlag, Vienna, p. 265-276, 2001.
- [27] F. Natterer, "Berechenbare Fehlerschranken für die Methode der finite Elemente," *International Series of Numerical Mathematics*, **28**, p. 109-121, Birkhäuser, 1975.
- [28] J.W.S. Rayleigh, The Theory of Sound, Vol. 1, Dover, 1945.
- [29] M.H. Schultz, Spline Analysis, Prentice-Hall, 1973.
- [30] G.L. Siganevich, "On the optimal estimation of error of the linear interpolation on a triangle of functions from $W_2^2(T)$ (in Russian)", *Doklady Akademii Nauk SSSR*, **300**(4), 811-814 (1988).
- [31] R. Temam, Numerical Analysis, D. Reidel Publishing Company, 1973.
- [32] E.T. Whittaker and G.N. Watson, *A Course of Modern Analysis*, 4th edn., Cambridge University Press, 1996
- [33] N. Yamamoto and N. Matsuda, "Validated computation for Bessel functions with multiple-precision (in Japanese)", *Transaction of Jap. Soc. for Indust. and Appl. Math.*, 15(3), 347-359 (2005).

A Determination of $\lambda^{(i)} = C_i^{-2}(+0)$ $(0 \le i \le 5)$

Recall Theorem 5 for the determination relations of $C_i(+0) := \lim_{\alpha \to +0} C_i(\alpha)$ or $\lambda^{(i)} = C_i^{-2}(+0)$ (i = 0, 1, 2, 3, 4, 5). Fortunately, all the ordinary differential equations (ODE) appearing there can be solved by means of the hypergeometric functions including the Bessel functions [32], so that we can obtain the determination equations as transcendental ones in terms of such functions. All the numerical results below are obtained by using Mathematica[®].

A.1 $\lambda^{(0)}$

From Theorem 5, the ODE and the boundary condition in this case are given by

$$-((1-s)u'(s))' = \lambda^{(0)}(1-s)u(s) \text{ for } s \in]0,1[, u'(0) = 0.$$
(157)

The general solution of the above ODE that can be identified with an element of $H^1(T) \supset V^{0,Z}$ is expressed by

$$u(s) = c^{(1)} J_0(\sqrt{\lambda^{(0)}}(1-s)), \qquad (158)$$

where $c^{(1)}$ is an arbitrary constant and J_0 is the 0-th order Bessel function of the first kind. Actually $Y_0(\sqrt{\lambda^{(0)}}(1-s))$ (Y_0 = the 0-th order Bessel function of the second kind) also satisfies the ODE but cannot be identified with an element of $H^1(T)$. Thus applying the boundary condition above and the relation $J_1 = -J'_0$, we have the following equation for $\lambda^{(0)}$:

$$J_1(\sqrt{\lambda^{(0)}}) = 0 , \qquad (159)$$

which means that $\sqrt{\lambda^{(0)}}$ is the smallest positive zero of J_1 . Thus we can obtain approximate values of $\lambda^{(0)}$ and $C_0(0+)$ as

$$\lambda^{(0)} \approx 3.83171^2$$
, $C_0(+0) \approx 0.260980$. (160)

A.2 $\lambda^{(1)} = \lambda^{(3)} = \lambda^{(4)}$

Similarly, the ODE, the linear constraint and the boundary condition in this case are given by (75) as

$$-((1-s)u'(s))' = \lambda^{(1)}(1-s)u(s) + C \text{ for } s \in]0,1[, \int_0^1 u(s)ds = 0, u'(0) = 0, (161)$$

where C is an arbitrary constant. Then the general solution of ODE that can be identified with an element of $H^1(T) \supset V^{1,Z}$ is expressed by

$$u(s) = c^{(1)} J_0(\sqrt{\lambda^{(1)}}(1-s)) - C(1-s) {}_1F_2(1;3/2,3/2;-\lambda^{(1)}(1-s)^2/4), \quad (162)$$

where $c^{(1)}$ is an arbitrary constant, and ${}_{1}F_{2}(\cdot; \cdot, \cdot, \cdot; \cdot)$ is a kind of hypergeometric function. Using the linear constraint and the boundary condition to the above, we have the following determination equation for $\lambda = \lambda^{(1)}$:

$$\frac{\lambda}{4} {}_{0}F_{1}(;2;-\frac{\lambda}{4}) {}_{2}F_{3}(1,1;\frac{3}{2},\frac{3}{2},2;-\frac{\lambda}{4}) + {}_{1}F_{2}(\frac{1}{2};1,\frac{3}{2};-\frac{\lambda}{4}) {}_{1}F_{2}(1;\frac{1}{2},\frac{3}{2};-\frac{\lambda}{4}) = 0, \quad (163)$$

where ${}_0F_1$ and ${}_2F_3$ are also hypergeometric functions. Solving this equation numerically, we have the following approximate values:

$$\lambda^{(1)} \approx 3.08126^2, \quad C_1(+0) \approx 0.324542.$$
 (164)

A.3 $\lambda^{(2)}$

By Theorem 5, the ODE and the boundary condition associated with $\lambda^{(2)}$ are given as

$$-((1-s)u'(s))' = \lambda^{(2)}(1-s)u(s) \text{ for } s \in]0,1[, u(0) = 0.$$
(165)

Then the general solution of the above ODE belonging to $H^1(T) \supset V^{2,Z}$ is the same as (158):

$$u(x) = c^{(1)} J_0(\sqrt{\lambda^{(2)}}(1-s))$$
(166)

so that the determination equation for $\lambda^{(2)}$ is obtained as

$$J_0(\sqrt{\lambda^{(2)}}) = 0.$$
 (167)

Thus $\sqrt{\lambda^{(2)}}$ is the minimum positive zero of J_0 . Approximately, we have

$$\lambda^{(2)} \approx 2.40483^2, \quad C_2(+0) \approx 0.415831.$$
 (168)

A.4 $\lambda^{(5)}$

By Theorem 5, the ODE and the boundary conditions associated to $\lambda^{(5)}$ are given as

$$((1-s)u''(s))'' = \lambda^{(5)}(1-s)u(s) \text{ for } s \in]0,1[, u(0) = u(1) = u''(0) = 0.$$
 (169)

Then the general solution of the ODE belonging to $H^2(T) \supset V^{4,Z}$ is

$$u(s) = c^{(1)} {}_{0}F_{3}(;\frac{1}{2},\frac{3}{4},\frac{3}{4};\frac{\lambda^{(5)}(1-s)^{4}}{256}) + c^{(2)}(1-s) {}_{0}F_{3}(;\frac{3}{4},1,\frac{5}{4};\frac{\lambda^{(5)}(1-s)^{4}}{256}) + c^{(3)}(1-s)^{2} {}_{0}F_{3}(;\frac{5}{4},\frac{5}{4},\frac{3}{2};\frac{\lambda^{(5)}(1-s)^{4}}{256}),$$
(170)

where $c^{(1)}$, $c^{(2)}$ and $c^{(3)}$ are arbitrary constants, and $_0F_3$ is a hypergeometric function. Then, introducing two functions $f(\lambda, t) = t_0F_3(; \frac{3}{4}, 1, \frac{5}{4}; \frac{\lambda}{256}t^4)$ and $g(\lambda, t) = t^2 {}_0F_3(; \frac{5}{4}, \frac{5}{4}, \frac{3}{2}; \frac{\lambda}{256}t^4)$, the determination equation for $\lambda = \lambda^{(5)}$ is given by

$$f''(\lambda, 1)g(\lambda, 1) - g''(\lambda, 1)f(\lambda, 1) = 0,$$
(171)

where $'' = \partial^2 / \partial t^2$. Approximately, we find that

$$\lambda^{(5)} \approx 9.26775^2, \quad C_5(+0) \approx 0.107901.$$
 (172)

Preprint Series, Graduate School of Mathematical Sciences, The University of Tokyo

UTMS

- 2008–9 Shouhei Ma: Fourier-Mukai partners of a K3 surface and the cusps of its Kähler Moduli.
- 2008–10 Takashi Tsuboi: On the uniform perfectness of diffeomorphism groups.
- 2008–11 Takashi Tsuboi: On the simplicity of the group of contactomorphisms.
- 2008–12 Hajime Fujita, Mikio Furuta and Takahiko Yoshida: Acyclic polarizations and localization of Riemann-Roch numbers I.
- 2008–13 Rolci Cipolatti and Masahiro Yamamoto: Inverse hyperbolic problem by a finite time of observations with arbitrary initial values.
- 2008–14 Yoshifumi Matsuda: Groups of real analytic diffeomorphisms of the circle with a finite image under the rotation number function.
- 2008–15 Shoichi Kaji: The (\mathfrak{g}, K) -module structures of the principal series representations of SL(4, R).
- 2008–16 G. Bayarmagnai: The (\mathfrak{g}, K) -module structures of principal series of SU(2,2).
- 2008–17 Takashi Tsuboi: On the group of real analytic diffeomorphisms.
- 2008–18 Takefumi Igarashi and Noriaki Umeda: Nonexistence of global solutions in time for reaction-diffusion systems with inhomogeneous terms in cones.
- 2008–19 Oleg Yu. Imanouilov, Gunther Uhlmann, and Masahiro Yamamoto: Partial data for the Calderón problem in two dimensions.
- 2008–20 Xuefeng Liu and Fumio Kikuchi: Analysis and estimation of error constants for P_0 and P_1 interpolations over triangular finite elements.

The Graduate School of Mathematical Sciences was established in the University of Tokyo in April, 1992. Formerly there were two departments of mathematics in the University of Tokyo: one in the Faculty of Science and the other in the College of Arts and Sciences. All faculty members of these two departments have moved to the new graduate school, as well as several members of the Department of Pure and Applied Sciences in the College of Arts and Sciences. In January, 1993, the preprint series of the former two departments of mathematics were unified as the Preprint Series of the Graduate School of Mathematical Sciences, The University of Tokyo. For the information about the preprint series, please write to the preprint series office.

ADDRESS:

Graduate School of Mathematical Sciences, The University of Tokyo 3–8–1 Komaba Meguro-ku, Tokyo 153-8914, JAPAN TEL +81-3-5465-7001 FAX +81-3-5465-7012